A MULTI-GROUP CONFIRMATORY FACTOR ANALYSIS OF THE REACTIVE-PROACTIVE AGGRESSION QUESTIONNAIRE IN A SAMPLE OF YOUNG ADULTS

A Thesis

Presented to

The Faculty of the Department of Criminal Justice and Criminology

Sam Houston State University

In Partial Fulfillment

of the Requirements for the Degree of

Master of Arts

by

Eric Meyers Cooke

December, 2016

A MULTI-GROUP CONFIRMATORY FACTOR ANALYSIS OF THE REACTIVE-PROACTIVE AGGRESSION QUESTIONNAIRE IN A SAMPLE OF YOUNG ADULTS

by

Eric Meyers Cooke

APPROVED:

Todd Armstrong, PhD Thesis Director

Danielle Boisvert, PhD Committee Member

Yan Zhang, PhD Committee Member

Phillip Lyons, PhD Dean, College of Criminal Justice

ABSTRACT

Cooke, Eric Meyers, A multi-group confirmatory factor analysis of the Reactive-Proactive Aggression questionnaire in a sample of young adults. Master of Arts (Criminal Justice and Criminology), December, 2016, Sam Houston State University, Huntsville, Texas.

Aggression has long-term negative effects on humanity as a whole. Because of this, aggression has become an important topic of study across many disciplines. Originally conceptualized as being either non-impulsive or impulsive, aggression has become dichotomized as being reactive or proactive. Each form of aggression has been linked to a variety of genetic, psychological, physiological, and neurobiological correlates. Because research continues to grow in these fields surrounding proactive and reactive aggression, it is important to make sure that measurement tools are assessing aggression appropriately across a variety of groups. One such tool that has emerged recently is the Reactive-Proactive Aggression Questionnaire (RPQ; Raine et al., 2006). Though a number of studies exist testing the validity of the RPQ in adolescent and child populations. No such studies exist examining the factor structure and measurement invariance of proactive and reactive aggression in young adult male and females from different ethnicities. The current study assesses factor structure and measurement invariance in multiple groups of young North American adults. Results show that a twofactor, proactive-reactive, structure fits the current data overall. However, measurement invariance is not achieved across the majority of these groups. Meaning that interpretation of reactive and proactive scores is not the same across gender and ethnicity. Limitations and future directions are discussed.

KEY WORDS: Aggression, Reactive, Proactive, Neurobiology, Genes, Maoa, 5-ht, Dopamine, Tryptophan, Amygdala, HPA axis, Frontal lobes, Confirmatory factor analysis

ACKNOWLEDGEMENTS

I would like to thank my committee chair, Todd Armstrong, PhD, as well as my committee members, Danielle Boisvert, PhD, and Yan Zhang, PhD. I would also like to thank Jessica Wells for all of her help and support through this process as well as all of the other graduate students that kept me going strong throughout all of this.

TABLE OF CONTENTS

		Page
ABSTI	RACT	iii
ACKN	IOWLEDGEMENTS	iv
TABL	E OF CONTENTS	v
LIST (OF TABLES	vii
CHAP'	TER	
I	INTRODUCTION	1
II	THE ETIOLOGY AND CAUSES OF PROACTIVE AND REACTIVE	
	AGGRESSION	8
	Heritability and Gene Studies	8
	Neurobiology	16
III	CURRENT STUDY	22
	Development and Validation of the RPQ	22
	Purpose and Research Questions	24
IV	METHODS	26
	Participants	26
	Measures	26
	Analytic Procedure	27
V	RESULTS	29
	Confirmatory Factor Analysis: Comparing the One- and Two-Factor Mode	els 29
	Multi-group Confirmatory Factor Analysis: Comparing Gender and Ethnic	eity 1

VI DISCUSSION	5
Limitations	7
Conclusion	8
REFERENCES	10
APPENDIX	36
VITA	37

LIST OF TABLES

Fable	Page
1	Demographics
2	Confirmatory Factor Analysis comparing overall fit of the one- and two-factor
	models without any questions removed or covariances created
3	Factor loadings comparing overall fit of the one- and two-factor models without
	any questions removed or covariances created
4	Confirmatory Factor Analysis comparing overall fit of the one- and two-factor
	models with questions removed and covariances created
5	Factor loadings comparing overall fit of the one- and two-factor models with
	questions removed and covariances created
6	MGCFA across gender2
7	MGCFA examining Caucasians and Hispanics
8	MGCFA examining Caucasians and the combined "Other" group

CHAPTER I

Introduction

Violent and aggressive acts have devastating and lasting impacts on the socioeconomic order of humanity. In 2014, it was estimated that 5.4 million non-fatal violent crimes occurred in the United States (Langton & Truman, 2014). Direct and indirect costs of violent crimes have been projected to exceed \$180 billion/year (McCollister, French, & Fang, 2010). Due to these overwhelmingly negative outcomes, aggressive behaviors are of great interest to scholars across a number of disciplines.

Historically, aggression related traits were dichotomized as either impulsive or non-impulsive (Barratt, Stanford, Dowdy, Liebman, & Kent, 1999). Impulsive aggression was characterized by a variety of neuropsychological correlates such as a lack of strategic processing (Stanford, Greve, & Gerstle, 1997), attentional difficulties, motor and cognitive impulsiveness, increased anger, and feelings of guilt after engaging in violent acts (Barratt et al., 1999). It was suggested that impulsive aggression stems from a dysfunctional stimuli-reward system, such that rewards for aggressive behavior become associated with stimuli that evoke aggressive responses, therefore creating a cycle where specific stimuli intensify aggressive acts (Berkowitz, 1974). Non-impulsive aggression, on the other hand, was associated with lower anxiety in general, dominance over others, and an increased desire for high social status (Barratt et al., 1999). Impulsive and non-impulsive aggression served as a good starting point to understanding the underlying traits of aggression. Recently, however, researchers have moved away from the

¹ Non-impulsive traits have also been referred to as premeditated (Barratt, Stanford, Dowdy, Liebman, & Kent, 1999) but the underlying characteristics are conceptually the same as those related to non-impulsive aggression.

characterization of aggression as being either impulsive or non-impulsive as the result of developing inconsistencies with the operationalization of both the impulsive and nonimpulsive facets (Meloy, 1992). These concepts were abandoned for more inclusive constructs, specifically, reactive and proactive aggression. As originally defined by Dodge (1991) and Meloy (1992), reactively aggressive individuals display emotional and uncontrolled violence in response to perceived physical or verbal aggression from others. Proactively aggressive individuals, on the other hand, lack emotion and use purposefully controlled aggressive acts to reach a desired goal (Dodge, 1991; Meloy, 1992). As previously mentioned, reactive aggression is a more encompassing concept than impulsive aggression. Though a core characteristic of reactive aggression is impulsivity (Chase, O'Leary, & Heyman, 2001; Kempes, Matthys, Vries, & Engeland, 2005), empirical research has found a wide range of psychological and physiological correlates to reactive aggression, including attentional difficulties and depression (Vitaro, Gendreau, Tremblay, & Oligny, 1998), high levels of peer victimization (Schwartz et al., 1998), social rejection (Price & Dodge, 1989), higher skin conductance and autonomic arousal (Hubbard et al., 2002), as well as higher levels of self-report anger and low levels of dominance (Chase et al., 2001). Like reactive aggression, proactive aggression is characterized by a series of psychological and physiological correlates such as high levels of dominance and instrumentality (Chase et al., 2001), low autonomic nervous system activity (Hubbard et al., 2002; Kempes et al., 2005), high levels of leadership and peer social status (Price & Dodge, 1989), the expectation of positive outcomes when using aggressive behavior (Schwartz et al., 1998), externalizing problems later in life (Vitaro, Brendgen, & Tremblay, 2002), and lower levels of anxiety (Vitaro et al., 1998).

Evidence shows that reactive and proactive aggression are distinct constructs. Various factor-analytic and meta-analytic studies have shown that both factors (reactive and proactive) are different but can also covary, to a certain degree, within individuals (Polman, Castro, Koops, Boxtel, & Merk, 2007; Poulin & Boivin, 2000; Raine et al., 2006; Salmivalli & Nieminen, 2002). If one were to imagine an aggression spectrum, reactive aggression would exist on one end and proactive aggression would exist on another, with specific characteristics from each overlapping in the middle. Because of the conceptual differences between proactive and reactive aggression, each has been grounded in its own theoretical framework.

Reactive aggression, which comes from the Frustration-Aggression Model (Dollard, Miller, Doob, Mowrer, & Sears, 1939), is speculated to occur when individuals view relatively ambiguous stimuli as hostile, and thus respond in a reactively aggressive manner (Dodge, Lochman, Harnish, Bates, & Pettit, 1997). Support for this comes from a variety of samples and experimental conditions (Atkins, Stoff, Osborne, & Brown, 1993; Crick & Dodge, 1996; Dodge et al., 1997; Hubbard et al., 2002; Quiggle, Garber, Panak, & Dodge, 1992; Scarpa & Raine, 1997). Thus suggesting, to some degree,² that reactive aggression is caused by individual perceptions of hostile stimuli.

Proactive aggression, on the other hand, has been framed within Social Learning Theory (Bandura, 1973). The basic idea here is that social information processing patterns lead to a step-wise decision making process that provides a possible means of obtaining a desired goal. Despite empirically driven attempts to test this idea (Atkins et

² Hostile attribution bias accounts for reactive aggression when psychological factors are included, as will be discussed in later sections, the specific etiology of reactive aggression is more complex than hostile attribution bias, alone, is able to explain.

al., 1993; Hubbard et al., 2002; Quiggle et al., 1992; Scarpa & Raine, 1997) little evidence has conclusively connected social information processing patterns to the expression of proactive aggression. This is most likely due to the instrumental nature of proactive aggression and the inability for standard social information processing measures to fully capture this nuanced behavior.³

Human behavior is dynamic and therefore, at times, can be hard to fully capture and separate. Proactive and reactive aggression are no different in this respect. Though they clearly reflect different behaviors (Polman et al., 2007; Poulin & Boivin, 2000; Raine et al., 2006; Salmivalli & Nieminen, 2002) there are certain characteristics of each aggression factor that correlate highly with the other. When examining the differences between proactive and reactive aggression, Raine and colleagues (2006) found that both forms of aggression were associated with excessive fighting at an early age, as well as stimulant seeking and paranoid ideation throughout adolescence. Similarly, Polman and colleagues (2007) meta-analytic study conclusively determined that proactive and reactive aggression can co-occur within individuals.

Further evidence for the comorbidity of these behaviors can be found in the overlap that characteristic traits of aggression share with various facets of psychopathy (Blair, 1999; Cornell et al., 1996; Frick, Cornell, Barry, Bodin, & Dane, 2003; Frick & Morris, 2004; Loney, Frick, Clements, Ellis, & Kerlin, 2003; Porter & Woodworth, 2006; Yildirim & Derksen, 2013). Specifically, the callous-unemotional (CU) traits that characterize Factor 1 psychopathy have been more strongly related to proactive/instrumental aggression, while reactive/impulsive aggressive behaviors

³ Similar to reactive aggression, the etiological pathways, to be discussed later, that lead to proactive aggression may explain its development in ways that theory alone cannot.

characterize Factor 2 psychopathic traits (Blair, 1999; Frick et al., 2003; Frick & Morris, 2004; Loney et al., 2003). There is strong evidence to suggest that a core feature of instrumental aggression is callous and unemotional behavioral traits. As previously mentioned, there is little evidence linking proactive aggression to social learning theory, in fact, there is more evidence linking CU traits to proactive aggression than to social learning theory. Because of this, moving forward in the next section, research will focus on the core characteristics of proactive aggression which will often be linked to the CU traits that describe Factor 1 psychopathy.

Because proactive and reactive aggression exist on a spectrum, it is extremely important to have reliable and valid aggression measurement tools. Measurement validity is arguably the focal point of scientific research. A variety of teacher, parental, and selfreport rating scales have been created to measure proactive and reactive aggression. One particular rating scale, which has emerged recently, is the Reactive-Proactive Aggression Questionnaire (RPQ; Raine et al., 2006). When developing the RPQ, Raine and colleagues (2006) first tested its measurement validity in a sample of 334 adolescent Caucasian and African-American 16 year old boys. After its development and subsequent release for scientific use, the RPQ was further validated in a variety of populations that included male and female Chinese schoolchildren (Fung, Raine, & Gao, 2009), male and female Italian adolescents (Fossati et al., 2009), and Dutch male and female adolescents (Cima, Raine, Meesters, & Popma, 2013). The studies mentioned show that the RPQ is a reliable and valid measure of aggression in child and adolescent populations across gender and culture. Recently, however, the RPQ has been used more widely across a variety of populations and experimental conditions that extend beyond the samples that

the above studies validated. Specifically, the RPQ has been used to assess proactive and reactive aggression in ethnically diverse, young adult, male and female samples (for examples see: Archer & Thanzami, 2009; Donnellan & Burt, 2015; Gao & Tang, 2013; Goodwin, Sellbom, & Salekin, 2015; Latzman, Vaidya, Clark, & Watson, 2011; Miller, Lynam, & Jones, 2008; Schenk, Fremouw, & Keelan, 2013). This poses a potential limitation because the factor structure and measurement invariance of the RPQ has only been tested in child and adolescent male and female samples with finite cultural differences. The current study seeks to address this limitation by testing the factor structure and measurement invariance of the Reactive-Proactive Aggression Questionnaire in an ethnically diverse sample of young adult males and females from North America.

To provide a basic overview, this paper will continue in three chapters. Chapter II discusses the etiological pathways and development of proactive and reactive aggression. The purpose of this chapter is to provide insight into a variety of genetic and neurobiological substrates that extend beyond theory to explain the onset of both forms of aggression. This chapter also aims to portray just how intricate proactive and reactive aggression are, and why there is a need to ensure that measurement tools, like the RPQ, are interpreted similarly across groups. Chapter III discusses the development of the RPQ and various studies that have validated it. This chapter will discuss limitations and measurement issues that the current study seeks to address. Chapter IV presents the methodology of the study. Participants, measurement, and analytic procedures are discussed within the framework of the provided data. Chapter V discusses the results,

with presentation of figures and tables. Chapter VI provides an overview of the findings with specific areas for future research, etiological meanings, and limitations.

CHAPTER II

The Etiology and Causes of Proactive and Reactive Aggression

Chapter II focuses on the etiological pathways and causes of proactive and reactive aggression. This extends beyond theory by providing specific genetic and neurobiological bases for the development of both forms of aggression. This section begins by examining the genetic heritability of aggression, which leads into a discussion of genetic polymorphisms that have been identified to explain aggression. Genetic effects on neurobiological factors are examined next, with emphasis placed on genetic-neurobiological pathways and gene-environment interactions. Neurobiological structures that potentially mediate the association between genetic variation and aggression subtypes will be included in this section to ensure a thorough review. It should be noted that this chapter provides a historical examination of the genetic and neurobiological factors that have been subject to the most scientific scrutiny. Other genetic and neurobiological explanations for aggression exist, however the topics discussed here are the most prominently researched and empirically sound.

Heritability and Gene Studies

Heritability. Heritability is a term used to describe how much variation in a specific phenotype, within a specific population, is due to genetic variation among individuals in that population. Meta-analysis of total aggression scores has found that genes, shared environmental factors, and non-shared environmental factors account for 44%, 6%, and 50% of the variance in aggression scores, respectively (Rhee & Waldman, 2002). Studies have also shown that genes account for roughly 38%-50% of variation in

⁴ This meta-analysis examined the one-factor aggression model rather than the two-factor (proactive-reactive) aggression paradigm.

reactive aggression (Baker, Raine, Liu, & Jacobson, 2008; Laubscher, Odendaal, Schneider, & Spies, 2013; Tuvblad & Baker, 2011)⁵. To the same end, non-shared environmental factors accounted for 40%-64%, while shared environmental factors accounted for 6%-15% of variation in reactive aggression (Baker et al., 2008; Laubscher et al., 2013; Tuvblad & Baker, 2011). When considering proactive aggression, genes account for 12%-50% of the variance, while non-shared environmental factors account for 40%-90% and shared environmental factors account for 6%-15% of the variance (Baker et al., 2008; Laubscher et al., 2013; Tuvblad & Baker, 2011). In short, there appears to be both a genetic and environmental influence on proactive and reactive aggression. However, heritability studies cannot point out which genes, specifically, influence the expression of both factors of aggression.

MAOA. Monoamine Oxidase A (MAOA) is a mitochondrial enzyme that catalyzes the oxidation of amines such as dopamine, norepinephrine, and serotonin. First identified by Sabol and Colleagues (1998), MAOA has been consistently linked to reactive/impulsive aggression. Researchers have found, across a wide range of samples and experimental conditions, that the low expression (L) allele of MAOA (MAOA-L) influences the onset and display of reactive aggression (Buckholtz & Meyer-Lindenberg, 2008; Fan, Fossella, Sommer, Wu, & Posner, 2003; Kuepper, Grant, Wielpuetz, & Hennig, 2013; Meyer-Lindenberg et al., 2006; Passamonti et al., 2006; Reif et al., 2007). Building off these findings, and paired with the knowledge that environmental factors influence reactive aggression (Baker et al., 2008; Laubscher et al., 2013; Tuvblad & Baker, 2011), researchers examined interactions between MAOA-L and various

⁵ Of all the studies cited in the section on heritability, only Baker et al. (2008) and Laubscher et al. (2013) utilized the RPQ as a measurement tool in their analysis.

environmental factors. For example, findings suggest that early life trauma interacts with the MAOA-L allele to influence the development of reactive aggression (Frazzetto et al., 2007). This interaction was also observed in males who reported a history of abuse (Huang et al., 2004). The high expression (H) allele of MAOA (MAOA-H) has been found to be associated with reactive aggression in females (Holz et al., 2016), while a gene by environment (G X E) interaction between MAOA-H and childhood physical abuse has predicted the onset of proactive aggression later in life in both males and females (Kolla, Attard, Craig, Blackwood, & Hodgins, 2014). Similarly, individuals carrying the 2 and 3 repeat variant alleles of the MAOA variable number tandem repeat (uVNTR) polymorphism are generally higher in impulsivity, while those carrying the 1 and 4 repeat variant alleles are lower in impulsivity (Manuck, Flory, Ferrell, Mann, & Muldoon, 2000).⁶

In all, MAOA-L seems to be related to reactive/impulsive aggression. Inconsistent evidence exists on the relation of MAOA-H to both reactive and proactive aggression. From a theoretical perspective, it has been proposed that individuals carrying the MAOA-L gene are not only reactively aggressive, but are presenting a hostile-attribution bias to ambiguous stimuli that results in a reactively aggressive response (Dodge, 2006). Currently, this model is purely hypothetical as no direct evidence has examined the relationship between hostile attribution bias, reactive aggression, and the MAOA-L gene. Future studies need to be conducted in order to tease out the exact relationship between these constructs.

. .

⁶ Null effects between MAOA and impulsivity have been noted by Skondras and colleagues (2004), but in this specific study, MAOA was measured by its presence in the platelet, which is not the best measure of MAOA.

Serotonin. The serotonergic system is one of the oldest amine systems of the brain. This system includes and consists of anything that pertains to or affects the neurotransmitter 5-HT. The serotonergic system includes a wide variety of transporter and receptor subtypes that innervate on most cortical and subcortical regions of the brain. 5-HT is linked to both reactive aggression (Booij et al., 2010; Coccaro, et al., 1997b; Coccaro, Fanning, Phan, & Lee, 2015; Davidson, Putnam, & Larson, 2000; de Boer & Koolhaas, 2005; Dolan & Anderson, 2003; Frankle et al., 2005; Homberg & Homberg, 2012; Jollant et al., 2007; Reif, et al., 2007; Siever et al., 1999; Soderstrom, Blennow, Manhem, & Forsman, 2001; van de Giessen et al., 2014) and proactive aggression (Davidson et al., 2000; de Boer & Koolhaas, 2005; Dolan & Anderson, 2003; Glenn, 2011; Homberg & Homberg, 2012; Soderstrom et al., 2001). In fact, over the decades that 5-HT has been examined, very few studies have found null effects between 5-HT and aggression (see Marseille, Lee, & Coccaro, 2012).

Genes influencing serotonergic function have long been associated with aggressive behaviors. For example, the 12/12 allele of the serotonergic-transporter variable number tandem repeat (5-HTT VNTR) polymorphism has strongly predicted the onset of reactive aggression (Aluja, Garcia, Blanch, De Lorenzo, & Fibla, 2009; Haberstick, Smolen, & Hewitt, 2006). However, due to 5-HTT VNTR being a relatively new discovery, replication of these findings needs to be conducted in other populations and experimental settings. Like the 12/12 allele of the 5-HTT VNTR polymorphism, the S allele of the serotonin-transporter-linked polymorphic region (5-HTTLPR S) has been related to reactive aggression (Aluja et al., 2009; Cadoret et al., 2003; Haberstick et al., 2006; Retz, Retz-Junginger, Supprian, Thome, & Rösler, 2004), while the L allele (5-

HTTLPR L) has been associated with proactive aggression (Aluja et al., 2009; Brummett, Siegler, Ashley-Koch, & Williams, 2011; Cadoret et al., 2003; Haberstick et al., 2006; McCaffery, Bleil, Pogue-Geile, Ferrell, & Manuck, 2003). From a theoretical standpoint, Sadeh and colleagues (2010) found that low socioeconomic status (SES) individuals homozygote for 5-HTTLPR L were more likely to display callous-unemotional and narcissistic traits that characterize proactive aggression. Keeping within the serotonergic transport system, van de Giessen and colleagues (2014) also found a positive correlation between excessive serotonergic transporter (5-HTT) binding and callous personality traits, thus furthering the evidence for a relationship between the serotonergic transport system, specific gene variants of said system, callous-unemotional traits, and proactive aggression.

Moving away from the serotonergic-transport system and its genetic polymorphisms, genes (HTR1B and HTR1A) coding for serotonergic inhibitory receptors (5-HT1B and 5-HT1A) have also been linked to aggression. Evidence shows that individuals carrying the GG genotype of HRT1A, which increases the expression of 5-HT1A (Benko et al., 2010) and the A-161T locus of HTR1B, which effects transcription of 5-HT1B (Zouk et al., 2007) are more reactively and impulsively aggressive. Similarly, increases in the density of serotonergic excitatory receptors (5-HT2A), which are influenced by the presence of the GC and C genotypes of HTR2A, have been linked to impulsive aggression (Giegling, Hartmann, Möller, & Rujescu, 2006). No research, to the authors knowledge, has yet to examine the relationship between these receptors and proactive aggression.

Though little direct evidence exists, it is worth mentioning that low levels of cerebrospinal fluid (CSF) 5-HIAA, the main metabolite of serotonin, has been linked to the onset and expression of reactive aggression (Coccaro & Lee, 2010; Linnoila et al., 1983), while normal levels of CSF 5-HIAA has been related to proactive aggression (Linnoila et al., 1983). This lends further support for the relationship between serotonin and aggression; however, to draw more conclusions from this more tests of CSF 5-HIAA needs to be conducted.

The serotonergic system, in short, plays a large role in the development and continual expression of reactive and proactive aggression. As mentioned, the serotonergic system innervates on a variety of cortical and subcortical regions of the brain. Later sections will discuss this interactive effect in more depth.

Tryptophan. Tryptophan hydroxylase (TPH) is the principal enzyme in the biosynthesis of serotonin (Zhang, Beaulieu, Sotnikova, Gainetdinov, & Caron, 2004). This enzyme converts L-tryptophan, an essential amino acid, into 5-HTP which is the precursor to the neurotransmitter serotonin. It has been found that carriers of the A218 C allele of the isozyme tryptophan hydroxylase (TPH2) were more likely to react impulsively when confronted with an ambiguous stimulus (Mercedes Perez-Rodriguez et al., 2010). This suggests that TPH2 is not only linked to impulsivity but also a hostile attribution style thought process. This idea was tested more thoroughly by Dougherty and colleagues (1999), who found that individuals with depleted levels of tryptophan were more likely to react in a hostile manner to the perceived change in behavior of others. Further support for this relationship comes from Crockett and colleagues (2008) who experimentally depleted the level of TPH in 20 participants. Manipulation of TPH in this

manner temporarily lowered serotonin levels in these individuals who, compared to matched placebo subjects, were more likely to react negatively to perceived unfair treatment by peers. Dougherty and colleagues (1999) and Crockett et al.'s (2008) experiments provide direct evidence of a hostile attribution bias (Crick & Dodge, 1996) in individuals with low levels of tryptophan. Though evidence is limited, it seems likely to suggest that tryptophan individually (Dougherty et al., 1999), or through an interactive effect on serotonin (Crockett et al., 2008), plays a role in reactive aggression and the display of a hostile-attribution bias in response to ambiguous stimuli.

Dopamine. Dopamine (DA) is a neurotransmitter that plays a major role in reward-motivated behavior. A variety of genes influence different dopaminergic transporter and receptor subtypes. The dopamine transporter (DAT1) has a rate limiting capacity on DA and is coded for by the gene SLC6A3. The 9 repeat allele of SLC6A3 has been linked to aggression in general (Young et al., 2002) and, more specifically, proactive aggression in violent subjects (Gerra et al., 2005). In contrast, Reif and colleagues (2007), through an examination of a male forensic population, found no association between the 9 repeat SLC6A3 allele and aggression

The genes DRD1, DRD2, DRD3, and DRD4, which code for dopamine receptors, have been suggested to influence aggressive behavior. In one of the earliest studies examining the effect that all four DRD genes had on aggression, Sweet et al. (1998) determined that only the DRD1 G polymorphism was associated with physical aggression. Conversely, Zai and colleagues (2012) found that the DRD2 polymorphism was associated with general aggression in children. Along the same lines, DA was examined within the context of a G X E interaction, where it was found that parental

maternal stress, paired with those presenting the DRD4 genotype, were higher in general aggression (Buchmann et al., 2014).

The final gene associated with aggression through its regulatory effect on DA is Catechol-O-methyl transferase (COMT). The COMT gene catalyzes the degradation of DA from the synaptic cleft. When considering reactive aggression, both the high activity Val158 (Perroud et al., 2010) and Met158 (Rujescu, Giegling, Gietl, Hartmann, & Möller, 2003) alleles have been associated with this form of aggression in adult suicide attempters. On the other hand, this same functional polymorphism of COMT has been associated with proactive aggression in schizophrenics (Tosato et al., 2011). Similarly, the COMT single nucleotide polymorphisms rs6269 and rs4818 have been linked to aggression in children, specifically in reference to callous-unemotional traits.

To summarize, evidence is somewhat mixed when considering genes that have a regulatory effect of dopamine and subsequent influence on aggression. There is evidence to suggest that dopamine plays a role in both proactive and reactive aggression, however, future research needs to focus on how exactly the same gene (such as COMT) can produce both proactive and reactive aggression. This could be due to the nuanced nature of aggression, but it could also be caused by the influence that MAOA has on DA. As previously mentioned, MAOA catabolizes various monoamines, including dopamine. Because of this, more research needs to examine whether aggression related traits are a direct effect of DA, or rather dysfunctionality within MAOA.

Hormones. There are a wide variety of hormones, but those most consistently studied in relation to proactive and reactive aggression are testosterone and cortisol. Results from the most conclusive studies of these hormones are mixed. For example,

some have found decreased cortisol to be related to reactive aggression (Cima, Smeets, & Jelicic, 2008; Feilhauer, Cima, Korebrits, & Nicolson, 2013; Poustka et al., 2010; Stadler et al., 2011)⁷. Others have found the opposite, suggesting that high levels of cortisol predict the expression of reactive aggression (Bokhoven et al., 2004; Denson, Mehta, & Ho Tan, 2013), especially in females displaying a hostile attribution bias in response to high and low levels of provocation (Böhnke, Bertsch, Kruk, Richter, & Naumann, 2010; Kobak, Zajac, & Levine, 2009). Still others propose that testosterone moderates cortisol (Popma et al., 2007) thus creating an interactive effect with 5-HT (Kuepper et al., 2010) which leads to the expression of reactive aggression. Taken together, the results of hormone studies and reactive aggression are not very consistent.

When considering proactive aggression, results of studies are less mixed which could be due to the fact that fewer studies have examined the effect that hormones have on proactive aggression (Yildirim & Derksen, 2013). Of the two studies that tested the relationship between hormones and traits leading to aggression, it was found that low cortisol levels in children (McBurnett, Lahey, Rathouz, & Loeber, 2000) and adolescents (Loney, Butler, Lima, Counts, & Eckel, 2006) were associated with callous-unemotional traits, which is a precursor to proactive aggression. This relationship was observed for cortisol only, not testosterone.

Neurobiology

HPA Axis. The hypothalamic-pituitary-adrenal (HPA) axis refers to a set of interconnected structures that interact and provide feedback among the three endocrine glands: the hypothalamus, the pituitary gland, and the adrenal glands. The HPA axis

⁷ Of all the studies cited in the section on hormones, only Feilhauer et al. (2013) used the RPQ as a measurement tool for their analysis.

constitutes a major part of the neuroendocrine system which controls, among other things, mood, emotions, and stress responsivity (Gotlib, Joormann, Minor, & Hallmayer, 2008). Due to its effect on stress-responsivity, the HPA axis has been linked to both forms of aggression (Meyer-Lindenberg et al., 2006).

Proactive forms of aggression have been linked to low (hypo) HPA activity (Dolan, Anderson, & Deakin, 2001; O'Leary, Taylor, & Eckel, 2010; Poustka et al., 2010; Stadler et al., 2011). This is most likely caused by the influence that genes have on the HPA axis which results in the display of CU traits and, ultimately, proactive aggression. Evidence for this comes from a meta-analytic study conducted by Miller et al. (2013), who found a significant association between the 5-HTTLPR genotype, such that the S allele displays dysfunctional cortisol activity which is directly related to HPA axis reactivity. Based on the relationship between cortisol and CU traits, Hawes and colleagues (2009) suggest that HPA axis hypoactivity heightens levels of CU traits. Ultimately, callous-unemotional traits would lead to the expression of proactive aggression as discussed by Frick et al. (2003).

HPA axis hyperreactivity has been linked to reactive and impulsive forms of aggression through the interaction between high cortisol levels and fear induced behavior (Lopez-Duran, Olson, Hajal, Felt, & Vazquez, 2008). Though this study revealed a strong relationship between cortisol, HPA axis reactivity, and reactive aggression, others have suggested that aggression is dependent on the ratio of cortisol and testosterone (Pavlov, Chistiakov, & Chekhonin, 2012). Like proactive aggression, a variety of genes that influence HPA axis activity have been linked to reactive aggression. These include the Arginine-Vasopression receptor gene 1B (AVPR1B; Luppino, Moul, Hawes, Brennan, &

Dadds, 2014), minor allele variants of the cholesterol transporter coding gene ABCG1 (Gietl et al., 2007), and the short GGC repeat allele of the gene that codes for androgen receptors (Rajender et al., 2008). Despite the influence that these genes have on the display of reactive aggression, they have not, as of yet, been linked to HPA axis hyperactivity.

Amygdala. The amygdala is an almond shaped cluster of nuclei located in the temporal lobe. The amygdala regulates emotions, emotional behavior, and motivation. Research has found that amygdalar hyporeactivity predicts proactive aggression (Blair, 2006; Deeley et al., 2006; Pardini, Raine, Erickson, & Loeber, 2014). Basically, this means that the less sensitive an amygdalar response is to stimuli (i.e. amygdalar underactivity), such as fear or anger in others faces, the more likely one is to act in a proactively aggressive manner. Research examining genetic effects on the amygdala, and the eventual display of proactive aggression, has supported the above interaction. Specifically, Hariri and colleagues (2002) found that the L allele of the 5-HTTLPR polymorphism was associated with amygdalar hypoactivity. This supports the hypothesized interaction between genetic influences on amygdalar hypoactivity and the development of callous-unemotional traits (Blair, 2006) which leads to the onset and display of proactive aggression (Frick et al., 2003). This interaction is further supported by Kiehl and colleagues (2001) and Marsh et al. (2008) who found that adolescents and adults displaying CU traits had reduced amygdala activity when presented with fearful stimuli.

Amygdalar hyperreactivity, on the other hand, has been linked to reactive aggression (Blair, 2006; Bobes et al., 2013; Carré, Fisher, Manuck, & Hariri, 2012;

Coccaro, McCloskey, Fitzgerald, & Phan, 2007; Montoya, Terburg, Bos, & Honk, 2011)⁸. In other words, individuals with an over active amygdala are more likely to respond reactively/impulsively to threat stimuli. The MAOA-L allele has been associated with amygdalar hyperactivity and the presentation of reactively aggressive traits in response to emotionally charged ques in multiple samples (Buckholtz & Meyer-Lindenberg, 2008; Meyer-Lindenberg et al., 2006) and adults specifically (Sebastian et al., 2010).

Frontal Lobe. The frontal lobe (forebrain) is one of the four major lobes of the cerebral cortex. It is made up of a variety of separate structures which dictate executive orders and decision making processes. Certain areas within the frontal lobe have been implicated in the development of aggression. For example, individuals presenting reactive and impulsive aggression have been found with damage, dysfunction, or lesions in the ventral medial prefrontal cortex (vmPFC; Anderson, Bechara, Damasio, Tranel, & Damasio, 1999; Bechara, Damasio, & Damasio, 2000; Bechara, Damasio, Damasio, & Lee, 1999; Bechara, Tranel, & Damasio, 2000; Blair, Mitchell, & Blair, 2005; Buchanan et al., 2010; Eslinger, 1998). Similarly, researchers have found decreases in the volume of the left orbitofrontal cortex compared to the right in reactively aggressive individuals (Antonucci et al., 2006; Blair, 2004; Gansler et al., 2009). This suggests that decreased brain matter in the frontal lobe causes reactively aggressive acts. Further support for this comes from Raine and Colleagues (1998) who found decreased volume in both the left and right prefrontal cortex in overtly aggressive murderers. Genes have also been found to influence certain areas of the frontal lobe with regards to reactive aggression.

.

⁸ Of all the studies cited in the section on the amygdala, only Bobes et al (2013) used the RPQ as a measurement tool in their analysis.

Eisenberger and colleagues (2007), for example, found that individuals carrying the MAOA-L allele showed hyperactivity in the dorsal Anterior Cingulate Cortex (dACC) which led to reactively aggressive traits in a social exclusion task.

Research suggests that proactively aggressive individuals display reduced amygdalar activity (Blair, 2004; Raine et al., 1998) and normal prefrontal activity (Raine et al., 1998) when compared to controls. Further assessment for the effect between genes and neurobiological substrates of proactive aggression needs to be examined before making any generalizations beyond implications of the above mentioned brain imaging studies.

The most salient idea is a full consideration of genetics, brain structures, and environmental factors, though few studies examine the interactive effect between all of these. Buckholtz and Meyer-Lindenberg (2008) found that carriers of the MAOA-L genotype exhibited excessive brain serotonin which led to amygdalar hyperreactivity and dysfunction in the medial prefrontal cortex and, ultimately, reactive aggression.

Similarly, Coccaro and Colleagues (2007) found that amygdalar dysfunction and decreased orbiotofrontal volume predicted the onset of reactive aggression. When considering proactive aggression, Finger and colleagues (2008) found that children with CU traits displayed dysfunctional activation of the vmPFC during a role reversal task.

Based on findings from Marsh and colleagues (2008) study, it seems that individuals with CU traits demonstrate reduced functional connectivity between the amygdala and vmPFC during a fear recognition task. Though the results of these studies are fairly conclusive, being the only studies of their kind, replication and extension needs to be done to fully

examine the interactive effect between various genes, brain structures, and environmental catalysts.

Collectively, these results show overlapping but distinguishable pathways for proactive and reactive aggression with clear evidence for genetic influences on neurobiological substrates. Despite the general consensus on research presented in this review, there is still some disagreement regarding the exact pathways that lead to the expression of both forms of aggression. Some of the ambiguity in this literature may stem from issues with the measurement of reactive and proactive aggression. Factor structure and measurement invariance of aggression measures, like the RPQ, have not been conclusively demonstrated across different groups. Some researchers continue to questions whether these measures generalize to adults, women, and different ethnic/racial groups. This is the current question that the current study seeks to address.

CHAPTER III

Current Study

Chapter III discusses the development and validation of the Reactive-Proactive Aggression Questionnaire (Raine et al., 2006). As previously mentioned, the RPQ was originally developed to capture the underlying traits of proactive and reactive aggression in an adolescent sample. More recently, the RPQ has been extended to a variety of populations, including young adults, males and females, and multiple ethnicities. This chapter will focus on the specific limitations in current validation studies of the RPQ and will end with an overview of the current research goals, questions, and analytic procedure.

Development and Validation of the RPQ

Raine and colleagues (2006) created the Reactive-Proactive Aggression

Questionnaire to address the lack of time efficient self-report measures that examined the differential constructs of proactive and reactive aggression. Development of the RPQ began after the collection of psychosocial and behavioral measures of 503 schoolboys (age 7). Once the participants reached age 16, they were given the first version of the RPQ, which included 26 questions. From there, exploratory factor analysis led to the deletion of three items, due to low endorsement frequencies and item-total correlations, and the final version of the RPQ, which is the 23 item scale used today. Confirmatory factor analysis of the RPQ (reported in Raine et al., 2006) yields a two-factor model of aggression that fits well to the data, however, these results are not without their limitations. As Raine and colleagues (2006) note, their sample was constricted to only male adolescents who were categorized as either Caucasian or African-American. These

sample limitations need to be addressed for three reasons. First, research shows that aggression is expressed differently depending on gender. Various studies suggest that males and females express reactive and proactive aggression differently. Marsee and Frick (2007), for example, found that females did not display a hostile attribution bias when engaging in reactive aggression. Also, female participants were more relationally proactively and reactively aggressive as opposed to males, who were more overtly aggressive. These findings are supported by others who consistently report gender differences in the characteristic expression of reactive and proactive aggression (Archer, 2000, 2004; Connor, Steingard, Anderson, & Jr, 2003; Murray-Close, Ostrov, Nelson, Crick, & Coccaro, 2010). Second, the RPQ was developed specifically for adolescents, however, recently it has been extended to college and adult populations. This poses a potential measurement issue because, as Kempes and colleagues (2005) suggest, age differences in the expression of reactive and proactive aggression exist. Specifically, it seems that older individuals are able to better control both forms of aggression in various situations. Third, factor structure and measurement of the RPQ was not examined across ethnicities other than Caucasian and African-American. It is important to take cultural differences into consideration when developing a psychometric measurement tool to ensure that there is no response bias across different cultures.

In order to validate the RPQ more fully, and address the limitations above, researchers have begun to examine the validity of the RPQ across gender, age, and ethnicity. Fossati and colleagues (2009) tested the validity of the RPQ in a large sample of Italian male and female high school students. Findings show that the RPQ was valid and reliable across ethnicity and gender. Along the same line, Fung et al. (2009) assessed

the RPQ in a large sample of male and female East Asian schoolchildren. Similar to Fossati et al.'s (2009) results, the psychometric properties of the RPQ were generalizable across ethnicity and gender. In the most comprehensive study to date, Cima and colleagues (2013) examined the validity of the RPQ in a large sample of Dutch male and females whose age ranged from 6 to 61 years. Results from the analysis demonstrate the continued replicability in the use of the RPQ in a variety of populations and age groups.

Results from these studies support the notion that the Reactive-Proactive Aggression Questionnaire is valid for use in a certain subset of samples. However, there is still some question as to the measurement invariance of the RPQ in multiple ethnic groups. Where Fossati et al. (2009) and Fung et al. (2009) assessed the RPQ in adolescent samples, the RPQ has been widely used to measure aggression in young adult samples (see Archer & Thanzami, 2009; Donnellan & Burt, 2015; Gao & Tang, 2013; Schenk, Fremouw, & Keelan, 2013). Further, differences in the validity of the RPQ across gender has only been assessed in adult Dutch populations (Cima et al., 2013), which could pose a potential problem when generalizing findings to older populations in North America. Finally, Cima et al. (2013), Fossati et al. (2009), and Fung et al. (2009) tested the ethnic differences of the RPQ in samples outside of the United States. Though the results were generalizable and valid in these populations, it speaks nothing towards the ethnic differences in the U.S. where the RPQ has been applied in samples of Caucasian and African Americans.

Purpose and Research Questions

In all, the RPQ is used quite widely but limited evidence exists for its measurement invariance in young adult samples. Similarly, no evidence exists of its

generalizability across race and gender groups in young adults within North America. The current study seeks to test the factor structure and measurement invariance of the RPQ in a diverse sample of young adults, while taking into account limitations and measurement considerations discussed above. The current study poses three research questions. First, will a two-factor (proactive-reactive) structure fit the current sample? Second, will equivalence of the model be achieved across gender? Finally, will equivalence of the model be achieved across ethnicity?

CHAPTER IV

Methods

Participants

The current sample (see Table 1) consisted of 237 male and 248 female (N= 485) students at a Southern University. Students who participated in this study were enrolled in a variety of undergraduate criminal justice courses. Mean age of participants was 21 years and ethnic differences were broken down as: 17.3% African American, 45.3% Caucasian, 27.6% Hispanic, 2.9% Asian, and 1.3% other. Analytic groups consisted of gender, with females as the reference group (mean male= .49, SD= .500) and ethnicity, with Caucasians as the reference group (mean Hispanic= .28, SD= .500; mean Other= .26, SD= .300). Due to sample size issues, African-Americans, Asians, and other ethnicities were combined into one group labeled "Other."

Measures

The Reactive-Proactive Aggression Questionnaire (RPQ; Raine et al., 2006; see Appendix A) is a 23 item, Likert-type scale that measures the two factor model of aggression, with 11 questions examining reactive aggression and 12 questions assessing proactive aggression. Questions ask participants "how often" they have done something, with response categories ranging from 0 (never), 1 (sometimes), or 2 (often). Example questions from the proactive side include "vandalized something for fun," and "yelled at others so they would do things for you." Example questions from the reactive portion include "reacted angrily when provoked by others," and "damaged things because you felt mad."

Analytic Procedure

The current study aims to test the factor structure of the RPQ across gender and ethnic groups. To begin with, total aggression (one-factor) scores were compared against proactive and reactive (two-factor) scores to ensure that a two-factor model fit the data better overall. Next, in order to make comparisons in regards to the factor structure across groups, a multi-group confirmatory factor analysis (MGCFA) was conducted. This type of analysis relies on tests of measurement invariance to examine whether survey items measure the same things across different groups (Cheung & Rensvold, 2002). Full model and multi-group analyses were conducted using the Robust Weighted Least Squares (WLSMV) estimator because it is robust to normality issues and small sample size.

In Mplus version 7.11 (Muthén & Muthén, 2012), MGCFA measurement invariance of ordered categorical data is tested by examining the difference between a sequence of increasingly restrictive models nested together (Brown, 2015). These models differ with regards to the constraints that are placed on factor loadings, thresholds, or both. Factor loadings illustrate the association between latent and observed variables while thresholds indicate the probability of selecting a particular response category. Muthén and Asparouhov (2002) and Brown (2015) recommend testing measurement invariance by, first, fitting a baseline (or free) model where factor loadings and thresholds are able to vary freely across groups. Second, testing this free parameter model to a model where factor loadings are constrained equal across groups and then to a model where factor loadings and thresholds are constrained to be equal across groups. The models were then nested within each other and compared using a variety of Goodness-of-Fit indices (see Brown, 2015; Cheung & Rensvold, 2002; Muthén et al., 2007;

Vandenberg & Lance, 2000) including chi-square difference test $(\Delta\chi^2)$, Root Mean Square Error of Approximation (RMSEA \leq .05), Comparative Fit Index (CFI \geq .9), and the Tucker-Lewis Index (TLI \geq .9).

Table 1

Demographics

	n	0/0
Male	237	48.9
Female	248	51.1
Caucasian	224	46.2
Hispanic	135	27.8
African American	84	17.3
Asian	14	2.9
Other	6	1.3

CHAPTER V

Results

Confirmatory Factor Analysis: Comparing the One- and Two-Factor Models

In order to determine overall model fit of the current data, a one-factor model was compared to a two-factor model. Goodness-of-fit indices and factor loadings are presented in tables 2 and 3, respectively. As can be seen, neither model fits particularly well to the data. The two-factor model (χ^2 = 1219.078, df= 229, CFI= .898, TLI= .887, RMSEA= .094) fits more adequately than the one-factor model (χ^2 = 1559.517, df= 230, CFI= .863, TLI= .849, RMSEA= .109), which is indicated by a chi-square difference ($\Delta\chi^2$ = 120, df= 1, p< .001). Though the two-factor model fits significantly better, neither model fits particularly well based on Goodness-of-fit indices discussed above, specifically RMSEA, CFI, and TLI.

Table 2

Confirmatory Factor Analysis comparing overall fit of the one- and two-factor models without any questions removed or covariances created

	χ^2	df	RMSEA	CFI	TLI	$\Delta \chi^2$, df, p
one-factor	1559.517	230	.109	.863	.849	
two-factor	1214.078	229	.094	.898	.887	120, 1, p<.001

Table 3

Factor loadings comparing overall fit of the one- and two-factor models without any questions removed or covariances created

	Pro	<u>active</u>	Reactive		
	One-factor	Two-factor	One-factor	Two-factor	
Yelled when annoyed			.332	.373	
Angry when provoked			.451	.492	
Temper tantrums			.243	.269	
Damaged things when mad			.374	.416	
Angry when frustrated			.385	.428	
Mad when lost game			.898	.911	
Angry when threatened			.375	.413	
Felt good after yelling or hitting			.483	.544	
Angry when lost game			.871	.893	
Hit to defend			.410	.451	
Angry when teased			.477	.535	
Had fights to be on top	.378	.478			
Taken things	.373	.459			
Vandalized for fun	.531	.628			
Had a gang fight to be cool	.568	.629			
Hurt others to win	.583	.689			
Used force to control others	.695	.794			
Threatened someone	.423	.507			
Used force to get money	.840	.882			
Made obscene phone calls	.339	.413			
Carried a weapon to fight	.390	.462			
Had others gang up on someone	.412	.501			
Yelled so others would do something for you	.350	.422			

Poor overall model fit led to the examination of modification indices (MI's) and factor loadings. This analysis led to the removal of reactive items five/seven as well as proactive items one/twelve as well as the covariation of reactive items nine with six, one with two, and one with three. After the removal and covariances of items, one- and two-

factor overall models were compared again. Goodness-of-fit indices and factor loadings are presented in tables 4 and 5. As can be seen, both the one-factor (χ^2 = 361.114, df= 149, CFI= .976, TLI= .972, RMSEA= .054) and two-factor (χ^2 = 264.476, df= 148, CFI= ..987, TLI= .985, RMSEA= .040) models fit better to the data overall. Chi-square difference test shows that the two-factor model fits the data significantly better than the one-factor model ($\Delta\chi^2$ = 30.155, df= 1, p<.001). Because of this, the two-factor model was used for multi-group comparisons.

Table 4

Confirmatory Factor Analysis comparing overall fit of the one- and two-factor models with questions removed and covariances created

	χ^2	df	RMSEA	CFI	TLI	$\Delta \chi^2$, df, p
one-factor	361.114	149	.054	.976	.972	
two-factor	264.476	148	.040	.987	.985	30.155, 1, p<.001

Table 5

Factor loadings comparing overall fit of the one- and two-factor models with questions removed and covariances created

	Proa	<u>ictive</u>	Reactive		
	One-factor	Two-factor	One-factor	Two-factor	
Yelled when annoyed			.154	.181	
Angry when provoked			.330	.370	
Temper tantrums			.228	.254	
Damaged things when mad			.458	.508	
Mad when lost game			.316	.354	
Felt good after yelling or hitting			.514	.585	
Angry when lost game			.302	.341	
Hit to defend			.434	.483	
Angry when teased continued			.519	.588	
Taken things	.455	.492			
Vandalized for fun	.615	.661			
Had a gang fight to be cool	.639	.660			
Hurt others to win	.662	.709			
Used force to control others	.762	.800			
Threatened someone	.478	.512			
Used force to get money	.883	.893			
Made obscene phone calls	.420	.454			
Carried a weapon to fight	.465	.500			
Had others gang up on someone	.487	.524			

Multi-group Confirmatory Factor Analysis: Comparing Gender and Ethnicity

When considering gender (see Table 6), the proactive-reactive subscales were more adequately fit to females (χ^2 = 202.261, df= 148, CFI= .989, TLI= .988, RMSEA= .039) than males (χ^2 = 266.945, df= 148, CFI= .959, TLI= .952, RMSEA= .057). Chi-square difference test of measurement invariance did not achieve invariance within gender when comparing free parameters to factor loading invariance ($\Delta\chi^2$ = 31.970, df=

15, p< .01) or to factor loading/threshold invariance ($\Delta \chi^2 = 97.508$, df= 34, p< .001). The same held true when examining factor loading invariance to factor loading/threshold invariance ($\Delta \chi^2 = 75.343$, df= 19, p<.001).

Table 6

MGCFA across gender

	χ^2	df	RMSEA	CFI	TLI	$\Delta \chi^2$, df, p
Females	202.261	148	.039	.989	.988	
Males	266.945	148	.057	.959	.952	
Free Parameters	381.621	296	.035	.988	.986	
Factor loading invariance	408.334	334	.036	.986	.985	
Factor loading/threshold invariance	467.505	330	.041	.981	.980	
Free parameters compared to factor loading invariance						31.970, 15, p< .01
Free parameters compared to factor loading/threshold invariance						87.508, 34, p< .001
Factor loading invariance compared to factor loading/threshold invariance						75.343, 19, p< .001

Table 7 reports results from the MGCFA when examining Caucasians compared to Hispanics. Adequate model fit was achieved for both Caucasians (χ^2 = 186.617, df= 148, CFI= .979, TLI= .976, RMSEA= .044) and Hispanics (χ^2 = 192.446, df= 148, CFI= .991, TLI= .990, RMSEA= .037). Chi-square difference test of measurement invariance

did achieve invariance between Caucasians and Hispanics when comparing free parameters to factor loading invariance ($\Delta\chi^2=23.193$, df= 15, p= .08). Measurement invariance, however, was not achieved between free parameters and factor loading/threshold invariance ($\Delta\chi^2=52.928$, df= 34, p< .05) as well as factor loading invariance compared to factor loading/threshold invariance ($\Delta\chi^2=37.719$, df= 19, p< .01).

Table 7

MGCFA examining Caucasians and Hispanics

						•
	χ^2	df	RMSEA	CFI	TLI	$\Delta \chi^2$, df, p
Caucasian	186.617	148	.044	.979	.976	
Hispanic	192.446	148	.037	.991	.990	
Free parameters	388.310	296	.042	.986	.984	
Factor loading invariance	405.541	311	.041	.986	.985	
Factor loading/threshold invariance	431.264	330	.041	.985	.985	
Free parameters compared to factor loading invariance						23.193, 15, p= .08
Free parameters compared to factor loading/threshold invariance						52.928, 34, p< .05
Factor loading invariance compared to factor loading/threshold invariance						37.719, 19, p< .01

Model fit for the combined ethnicity of the Other group (see table 8) also adequately fit the data (χ^2 = 195.133, df= 148, CFI= .987, TLI= .985, RMSEA= .050).

Compared to Caucasians, chi-square difference test of measurement invariance was achieved when testing free parameters to factor loading invariance ($\Delta\chi^2$ = 21.814, df= 15, p= .113). But measurement invariance was not achieved when examining free parameters to factor loading/threshold invariance ($\Delta\chi^2$ = 72.776, df= 34, p< .001) as well as factor loading invariance to factor loading/threshold invariance ($\Delta\chi^2$ = 59.274, df= 19, p< .001).

Table 8

MGCFA examining Caucasians and the combined "Other" group

	χ^2	df	RMSEA	CFI	TLI	$\Delta \chi^2$, df, p
Caucasian	186.617	148	.044	.979	.976	
Other	195.133	148	.050	.987	.985	
Free parameters	418.084	296	.049	.985	.983	
Factor loading invariance	429.701	311	.047	.986	.984	
Factor loading/threshold invariance	477.376	330	.051	.982	.982	
Free parameters compared to factor loading invariance						21.814, 15, p= .113
Free parameters compared to factor loading/threshold invariance						72.776, 34, p< .001
Factor loading invariance compared to factor loading/threshold invariance						59.274, 19, p< .001

CHAPTER VI

Discussion

This study highlights the importance of the continuous need to examine measurement differences of survey tools across multiple groups. Though the RPQ has been validated in a variety of samples, the current findings suggest that groups, specifically gender and ethnicity, can differ in their variance of responses to survey questions. With the finding that measurement invariance can differ across groups, based on the constraints placed on different parameters, this study hopes to successfully show that measurement issues should be continuously examined when using survey tools in diverse populations.

The objective of the current study was three-fold. First, the current study sought to determine whether or not a two-factor, proactive and reactive, structure of the RPQ fit the current sample. Second, to test the equivalence of the reactive-proactive model across gender. And third, to test the equivalence of the reactive-proactive model across ethnicity. Results from the analysis suggest that a two-factor model of proactive and reactive aggression fits the current sample significantly better than a one-factor, or general aggression, model. This two-factor model was then used to examine group differences between gender and ethnicity. Findings suggest that the two-factor model fits well for both males and females, though there were significant difference in survey measurement between the two groups. This lack of invariance means that the scores of males and females, on the RPQ, cannot be interpreted the same. Comparison of Caucasians to Hispanics, however, shows measurement invariance only when factor loading constraints are imposed. When stricter constraints are enforced (i.e. factor

loadings and thresholds are constrained) measurement invariance is lost, meaning that scores between these groups cannot be interpreted the same. These results held true for the combined "Other" group compared to Caucasians. It seems that model fit was achieved for the group itself, but a difference existed in survey measurement results compared to Caucasians when parameters were constrained. Future research should examine why, exactly, imposing stricter parameterizations leads to the difference in invariance as well as what questions are conceptualized differently between the various groups.

Findings that the two-factor structure fit the data better than the one-factor model support recent validation studies of the RPQ conducted by Raine et al. (2006), Cima and colleagues (2013), Fung, Raine, and Gao (2009), and Fossatei et al. (2009). Results from this study add to the continuously growing research suggesting that proactive and reactive aggression is a full and more encompassing way to think about aggression than a general model. Results from this study, however, were surprising in regards to the level of modification that needed to be done to the RPQ before the authors were able to achieve significant fit to the data. It appears that none of the other RPQ validation studies had to remove items from the analysis to achieve overall fit. Not only were four items removed from this study, but covariances were created between three items. It is also surprising that the sample size of the current study, N= 485, was larger than that of Raine et al.'s (2006) original study, N= 335, where the RPQ was developed. Though this could be due to the mixed gender/ethnic non-clinical population of the current study, future factor analytic studies of the RPQ should take this into account and, perhaps, consider

examining the structure of the RPQ under a more rigorous analytic procedure, such as Item Response Theory.

The discovery that there were significant differences in measurement across gender is not surprising. It seems that research showing differences in the expression of aggression across gender (Archer, 2000; Connor, Steingard, Anderson, & Melloni Jr, 2003; Marsee & Frick, 2007; Murray-Close, Ostrov, Nelson, Crick, & Coccaro, 2010) can be related to differences in response scores on the RPQ between males and females. More research needs to be conducted to determine why variance existed between these groups, and what questions they differed on, in order to develop a more encompassing measure of aggression. In regards to ethnicity, results do not seem to support previous analysis of the RPQ across culture (Raine et al., 2006; Cima et al., 2013; Fung, Raine, and Gao, 2009; and Fossatei et al., 2009). There does seem to be a significant difference in the RPQ's measurement of proactive and reactive aggression between Caucasians and Hispanics as well as Caucasians and Other groups in young adults. Future research should test this finding more thoroughly as it is important to know that items are measuring the same concepts across a wide variety of ethnic groups.

Limitations

The current study was limited in several ways. First, due to the small sample size it was impossible to test measurement invariance in homogenous subgroups. Specifically, there was no way to examine detailed gender-ethnic interactions that might influence the measurement of proactive and reactive aggression within groups. Future research should try to examine subsets of groups (for example Hispanic males) within the larger group paradigm.

Second, due to sample size issues with ethnic groups, we are still unsure as to the generalizability and measurement invariance between other ethnic groups. As mentioned previously, the analysis compared Caucasians to a combined "Other" group. This group consisted of African-Americans, Asians, and other categories. Though there was a difference in measurement between this group and the Caucasian group, it is nearly impossible to tell whether this is caused by one specific ethnic group, or even if this finding is generalizable beyond this study.

Third, levels of aggression in college populations is lower than forensic populations overall. Because of this, it is difficult to determine if these results are generalizable to samples that are at high risk of displaying both proactive and reactive aggression. It would be interesting if, in the future, one were able to examine the factor structure of the RPQ in both college and forensic populations to determine if the group differences found in this study persist.

Fourth, because sample size was limited, the current study employed the WLSMV estimator. This could cause differences in results when compared to previous studies examining the RPQ. It is important for future research to re-test these findings in a larger sample in order to avoid issues with the estimator.

Conclusion

Results from this study show that the two-factor model describes aggression much better than the one-factor structure. There do seem to be significant differences in the interpretation of the RPQ across gender and ethnic groups in a young adult (non-forensic) population. This is quite important as it shows that the etiology of aggression is being measured differently across groups. When looking back to the etiology and causes of

proactive and reactive aggression, it is important to have measurement tools that consistently measure similar constructs in different groups. Future analyses should be conducted to determine why the RPQ assesses the intricacies of proactive and reactive aggression differently across gender and ethnic groups.

REFERENCES

- Aluja, A., Garcia, L. F., Blanch, A., De Lorenzo, D., & Fibla, J. (2009). Impulsive-disinhibited personality and serotonin transporter gene polymorphisms:

 Association study in an inmate's sample. Journal of Psychiatric Research, 43(10), 906–914. http://doi.org/10.1016/j.jpsychires.2008.11.008
- Anderson, S. W., Bechara, A., Damasio, H., Tranel, D., & Damasio, A. R. (1999).

 Impairment of social and moral behavior related to early damage in human prefrontal cortex. Nature Neuroscience, 2(11), 1032.
- Antonucci, A. S., Gansler, D. A., Tan, S., Bhadelia, R., Patz, S., & Fulwiler, C. (2006).

 Orbitofrontal correlates of aggression and impulsivity in psychiatric patients.

 Psychiatry Research: Neuroimaging, 147(2–3), 213–220.

 http://doi.org/10.1016/j.pscychresns.2005.05.016
- Archer, J. (2000). Sex differences in aggression between heterosexual partners: A meta-analytic review. Psychological Bulletin, 126(5), 651–680. http://doi.org/10.1037/0033-2909.126.5.651
- Archer, J. (2004). Sex Differences in Aggression in Real-World Settings: A Meta-Analytic Review. Review of General Psychology, 8(4), 291–322. http://doi.org/10.1037/1089-2680.8.4.291
- Archer, J., & Thanzami, V. (2009). The relation between mate value, entitlement, physical aggression, size and strength among a sample of young Indian men. Evolution and Human Behavior, 30(5), 315–321.

- Atkins, M. S., Stoff, D. M., Osborne, M. L., & Brown, K. (1993). Distinguishing
 Instrumental and Hostile Aggression: Does It Make a Difference? Journal of
 Abnormal Child Psychology, 21(4), 355–365.
- Baker, L. A., Raine, A., Liu, J., & Jacobson, K. C. (2008). Differential Genetic and Environmental Influences on Reactive and Proactive Aggression in Children.
 Journal of Abnormal Child Psychology, 36(8), 1265–1278.
 http://doi.org/10.1007/s10802-008-9249-1
- Bandura, A. (1973). Aggression: A social learning analysis. Prentice-Hall. Retrieved from http://psycnet.apa.org/psycinfo/1974-00914-000
- Barratt, E. S., Stanford, M. S., Dowdy, L., Liebman, M. J., & Kent, T. A. (1999).

 Impulsive and premeditated aggression: a factor analysis of self-reported acts.

 Psychiatry Research, 86(2), 163–173. http://doi.org/10.1016/S01651781(99)00024-4
- Bechara, A., Damasio, H., & Damasio, A. R. (2000). Emotion, Decision Making and the Orbitofrontal Cortex. Cerebral Cortex, 10(3), 295–307. http://doi.org/10.1093/cercor/10.3.295
- Bechara, A., Damasio, H., Damasio, A. R., & Lee, G. P. (1999). Different Contributions of the Human Amygdala and Ventromedial Prefrontal Cortex to Decision-Making. The Journal of Neuroscience, 19(13), 5473–5481.
- Bechara, A., Tranel, D., & Damasio, H. (2000). Characterization of the decision-making deficit of patients with ventromedial prefrontal cortex lesions. Brain, 123(11), 2189–2202. http://doi.org/10.1093/brain/123.11.2189

- Benko, A., Lazary, J., Molnar, E., Gonda, X., Tothfalusi, L., Pap, D., ... Bagdy, G.
 (2010). Significant association between the C(-1019)G functional polymorphism of the HTR1A gene and impulsivity. American Journal of Medical Genetics Part
 B: Neuropsychiatric Genetics, 153B(2), 592-599.
 http://doi.org/10.1002/ajmg.b.31025
- Berkowitz, L. (1974). Some determinants of impulsive aggression: Role of mediated associations with reinforcements for aggression. Psychological Review, 81(2), 165–176. http://doi.org/10.1037/h0036094
- Blair. (2004). The roles of orbital frontal cortex in the modulation of antisocial behavior.

 Brain and Cognition, 55(1), 198–208. http://doi.org/10.1016/S02782626(03)00276-8
- Blair, J., Mitchell, D., & Blair, K. (2005). The Psychopath: emotion and the Brain.

 Blackwell Publishing. Retrieved from http://psycnet.apa.org/psycinfo/2005
 11002-000
- Blair, R. J. R. (1999). Responsiveness to distress cues in the child with psychopathic tendencies. Personality and Individual Differences, 27(1), 135–145.
- Blair, R. J. R. (2006). The emergence of psychopathy: Implications for the neuropsychological approach to developmental disorders. Cognition, 101(2), 414–442. http://doi.org/10.1016/j.cognition.2006.04.005
- Bobes, M. A., Ostrosky, F., Diaz, K., Romero, C., Borja, K., Santos, Y., & Valdés-Sosa, M. (2013). Linkage of functional and structural anomalies in the left amygdala of reactive-aggressive men. Social Cognitive and Affective Neuroscience, 8(8), 928–936. http://doi.org/10.1093/scan/nss101

- Böhnke, R., Bertsch, K., Kruk, M. R., Richter, S., & Naumann, E. (2010). Exogenous cortisol enhances aggressive behavior in females, but not in males.

 Psychoneuroendocrinology, 35(7), 1034–1044.

 http://doi.org/10.1016/j.psyneuen.2010.01.004
- Bokhoven, I. van, Goozen, S. H. M. V., Engeland, H. van, Schaal, B., Arseneault, L.,
 Séguin, J. R., ... Tremblay, R. E. (2004). Salivary cortisol and aggression in a
 population-based longitudinal study of adolescent males. Journal of Neural
 Transmission, 112(8), 1083–1096. http://doi.org/10.1007/s00702-004-0253-5
- Booij, L., Tremblay, R. E., Leyton, M., Séguin, J. R., Vitaro, F., Gravel, P., ... Benkelfat,
 C. (2010). Brain Serotonin Synthesis in Adult Males Characterized by Physical
 Aggression during Childhood: A 21-Year Longitudinal Study: e11255. PLoS
 One, 5(6). http://doi.org/http://dx.doi.org/10.1371/journal.pone.0011255
- Bribaumer, N., & others. (2005). Deficient fear conditioning in psychopathy. Archives of General Psychiatry, 62, 799–805.
- Brown, T. A. (2015). Confirmatory factor analysis for applied research. Guilford

 Publications. Retrieved from

 https://books.google.com/books?hl=en&lr=&id=tTL2BQAAQBAJ&oi=fnd&pg=

 PP1&dq=confirmatory+factor+analysis+for+applied+research+AND+brown&ots

 =ajTzrPWL8D&sig=6odmII5vqnPHaAuiQdcFtjvTh9o
- Brummett, B. H., Siegler, I. C., Ashley-Koch, A., & Williams, R. B. (2011). Effects of 5HTTLPR on cardiovascular response to an emotional stressor. Psychosomatic Medicine, 73(4), 318–322. http://doi.org/10.1097/PSY.0b013e3182118c16

- Buchanan, T. W., Driscoll, D., Mowrer, S. M., Sollers III, J. J., Thayer, J. F., Kirschbaum, C., & Tranel, D. (2010). Medial prefrontal cortex damage affects physiological and psychological stress responses differently in men and women. Psychoneuroendocrinology, 35(1), 56–66. http://doi.org/10.1016/j.psyneuen.2009.09.006
- Buchmann, A. F., Zohsel, K., Blomeyer, D., Hohm, E., Hohmann, S., Jennen-steinmetz,
 C., ... Laucht, M. (2014). Interaction between prenatal stress and dopamine D4 receptor genotype in predicting aggression and cortisol levels in young adults.
 Psychopharmacology, 231(16), 3089–97.
 http://doi.org/http://dx.doi.org/10.1007/s00213-014-3484-7
- Buckholtz, J. W., & Meyer-Lindenberg, A. (2008). MAOA and the neurogenetic architecture of human aggression. Trends in Neurosciences, 31(3), 120–129. http://doi.org/10.1016/j.tins.2007.12.006
- Cadoret, R. J., Langbehn, D., Caspers, K., Troughton, E. P., Yucuis, R., Sandhu, H. K., & Philibert, R. (2003). Associations of the serotonin transporter promoter polymorphism with aggressivity, attention deficit, and conduct disorder in an adoptee population. Comprehensive Psychiatry, 44(2), 88–101. http://doi.org/http://dx.doi.org/10.1053/comp.2003.50018
- Carré, J. M., Fisher, P. M., Manuck, S. B., & Hariri, A. R. (2012). Interaction between trait anxiety and trait anger predict amygdala reactivity to angry facial expressions in men but not women. Social Cognitive and Affective Neuroscience, 7(2), 213–221. http://doi.org/10.1093/scan/nsq101

- Chase, K. A., O'Leary, K. D., & Heyman, R. E. (2001). Categorizing partner-violent men within the reactive–proactive typology model. Journal of Consulting and Clinical Psychology, 69(3), 567–572. http://doi.org/10.1037/0022-006X.69.3.567
- Cheung, G. W., & Rensvold, R. B. (2002). Evaluating goodness-of-fit indexes for testing measurement invariance. Structural Equation Modeling, 9(2), 233–255.
- Cima, M., Raine, A., Meesters, C., & Popma, A. (2013). Validation of the Dutch

 Reactive Proactive Questionnaire (RPQ): differential correlates of reactive and

 proactive aggression from childhood to adulthood. Aggressive Behavior, 39(2),

 99–113.
- Cima, M., Smeets, T., & Jelicic, M. (2008). Self-reported trauma, cortisol levels, and aggression in psychopathic and non-psychopathic prison inmates. Biological Psychology, 78(1), 75–86. http://doi.org/10.1016/j.biopsycho.2007.12.011
- Coccaro, E. F., Fanning, J. R., Phan, K. L., & Lee, R. (2015). Serotonin and impulsive aggression. CNS Spectrums, 20(03), 295–302.
- Coccaro, E. F., Kavoussi, R. J., Sheline, Y. I., Berman, M. E., & Csernansky, J. G. (1997a). Impulsive Aggression in Personality Disorder Correlates with Platelet 5-HT2A Receptor Binding. Neuropsychopharmacology, 16(3), 211–216. http://doi.org/10.1016/S0893-133X(96)00194-7
- Coccaro, E. F., Kavoussi, R. J., Trestman, R. L., Gabriel, S. M., Cooper, T. B., & Siever, L. J. (1997b). Serotonin function in human subjects: intercorrelations among central 5-HT indices and aggressiveness. Psychiatry Research, 73(1–2), 1–14. http://doi.org/10.1016/S0165-1781(97)00108-X

- Coccaro, E. F., & Lee, R. (2010). Cerebrospinal fluid 5-hydroxyindolacetic acid and homovanillic acid: reciprocal relationships with impulsive aggression in human subjects. Journal of Neural Transmission, 117(2), 241–248. http://doi.org/10.1007/s00702-009-0359-x
- Coccaro, E. F., McCloskey, M. S., Fitzgerald, D. A., & Phan, K. L. (2007). Amygdala and Orbitofrontal Reactivity to Social Threat in Individuals with Impulsive Aggression. Biological Psychiatry, 62(2), 168–178. http://doi.org/10.1016/j.biopsych.2006.08.024
- Connor, D. F., Steingard, R. J., Anderson, J. J., & Jr, R. H. M. (2003). Gender

 Differences in Reactive and Proactive Aggression. Child Psychiatry and Human

 Development, 33(4), 279–294. http://doi.org/10.1023/A:1023084112561
- Cornell, D. G., Warren, J., Hawk, G., Stafford, E., Oram, G., & Pine, D. (1996).

 Psychopathy in instrumental and reactive violent offenders. Journal of Consulting and Clinical Psychology, 64(4), 783–790. http://doi.org/10.1037/0022-006X.64.4.783
- Crick, N. R., & Dodge, K. A. (1996). Social information-processing mechanisms in reactive and proactive aggression. Child Development, 67(3), 993–1002.
- Crockett, M. J., Clark, L., Tabibnia, G., Lieberman, M. D., & Robbins, T. W. (2008).

 Serotonin modulates behavioral reactions to unfairness. Science (New York, N.Y.), 320(5884), 1739. http://doi.org/10.1126/science.1155577
- Davidson, R. J., Putnam, K. M., & Larson, C. L. (2000). Dysfunction in the Neural Circuitry of Emotion Regulation: A Possible Prelude to Violence. Science, 289(5479), 591–594.

- de Boer, S. F., & Koolhaas, J. M. (2005). 5-HT1A and 5-HT1B receptor agonists and aggression: A pharmacological challenge of the serotonin deficiency hypothesis. European Journal of Pharmacology, 526(1–3), 125–139. http://doi.org/10.1016/j.ejphar.2005.09.065
- Deeley, Q., Daly, E., Surguladze, S., Tunstall, N., Mezey, G., Beer, D., ... Murphy, D. G. (2006). Facial emotion processing in criminal psychopathy. The British Journal of Psychiatry, 189(6), 533–539. http://doi.org/10.1192/bjp.bp.106.021410
- Denson, T. F., Mehta, P. H., & Ho Tan, D. (2013). Endogenous testosterone and cortisol jointly influence reactive aggression in women. Psychoneuroendocrinology, 38(3), 416–424. http://doi.org/10.1016/j.psyneuen.2012.07.003
- Dodge, K. A. (1991). The structure and function of reactive and proactive aggression. In D. J. Pepler & K. H. Rubin (Eds.), The development and treatment of childhood aggression (pp. 201–218). Hillsdale, NJ, England: Lawrence Erlbaum Associates, Inc.
- Dodge, K. A. (2006). Translational science in action: Hostile attributional style and the development of aggressive behavior problems. Development and Psychopathology, 18(03), 791–814.
- Dodge, K. A., Lochman, J. E., Harnish, J. D., Bates, J. E., & Pettit, G. S. (1997). Reactive and proactive aggression in school children and psychiatrically impaired chronically assaultive youth. Journal of Abnormal Psychology, 106(1), 37–51. http://doi.org/10.1037/0021-843X.106.1.37
- Dolan, M., Anderson, I. M., & Deakin, J. F. W. (2001). Relationship between 5-HT function and impulsivity and aggression in male offenders with personality

- disorders. The British Journal of Psychiatry, 178(4), 352–359. http://doi.org/10.1192/bjp.178.4.352
- Dolan, M. C., & Anderson, I. M. (2003). The Relationship between Serotonergic Function and the Psychopathy Checklist: Screening Version. Journal of Psychopharmacology, 17(2), 216–222. http://doi.org/10.1177/0269881103017002011
- Dollard, J., Miller, N. E., Doob, L. W., Mowrer, O. H., & Sears, R. R. (1939). Frustration and aggression. Retrieved from http://doi.apa.org/psycinfo/2004-16227-000
- Donnellan, M. B., & Burt, S. A. (2015). A Further Evaluation of the Triarchic Conceptualization of Psychopathy in College Students. Journal of Psychopathology and Behavioral Assessment, 1–11.
- Dougherty, D. M., Bjork, J. M., Marsh, D. M., & Moeller, F. G. (1999). Influence of trait hostility on tryptophan depletion-induced laboratory aggression. Psychiatry Research, 88(3), 227–232. http://doi.org/10.1016/S0165-1781(99)00088-8
- Eisenberger, N. I., Way, B. M., Taylor, S. E., Welch, W. T., & Lieberman, M. D. (2007). Understanding genetic risk for aggression: clues from the brain's response to social exclusion. Biological Psychiatry, 61(9), 1100–1108.
- Eslinger, P. J. (1998). Neurological and neuropsychological bases of empathy. European Neurology, 39(4), 193–199.
- Fan, J., Fossella, J., Sommer, T., Wu, Y., & Posner, M. I. (2003). Mapping the Genetic Variation of Executive Attention onto Brain Activity. Proceedings of the National Academy of Sciences of the United States of America, 100(12), 7406–7411.

- Feilhauer, J., Cima, M., Korebrits, A., & Nicolson, N. A. (2013). Salivary cortisol and psychopathy dimensions in detained antisocial adolescents.
 Psychoneuroendocrinology, 38(9), 1586–1595.
 http://doi.org/10.1016/j.psyneuen.2013.01.005
- Finger, E. C., Marsh, A. A., Mitchell, D. G., Reid, M. E., Sims, C., Budhani, S., ... others. (2008). Abnormal ventromedial prefrontal cortex function in children with psychopathic traits during reversal learning. Archives of General Psychiatry, 65(5), 586–594.
- Fossati, A., Raine, A., Borroni, S., Bizzozero, A., Volpi, E., Santalucia, I., & Maffei, C. (2009). A cross-cultural study of the psychometric properties of the Reactive—

 Proactive Aggression Questionnaire among Italian nonclinical adolescents.

 Psychological Assessment, 21(1), 131–135. http://doi.org/10.1037/a0014743
- Frankle, W. G., Lombardo, I., New, A. S., Goodman, M., Talbot, P. S., Huang, Y., ... others. (2005). Brain serotonin transporter distribution in subjects with impulsive aggressivity: a positron emission study with [11C] McN 5652. American Journal of Psychiatry, 162(5), 915–923.
- Frazzetto, G., Lorenzo, G. D., Carola, V., Proietti, L., Sokolowska, E., Siracusano, A., ...

 Troisi, A. (2007). Early Trauma and Increased Risk for Physical Aggression

 during Adulthood: The Moderating Role of MAOA Genotype: e486. PLoS One,

 2(5). http://doi.org/http://dx.doi.org/10.1371/journal.pone.0000486
- Frick, P. J., Cornell, A. H., Barry, C. T., Bodin, S. D., & Dane, H. E. (2003). Callousunemotional traits and conduct problems in the prediction of conduct problem

- severity, aggression, and self-report of delinquency. Journal of Abnormal Child Psychology, 31(4), 457–470.
- Frick, P. J., & Morris, A. S. (2004). Temperament and developmental pathways to conduct problems. Journal of Clinical Child and Adolescent Psychology, 33(1), 54–68.
- Fung, A. L.-C., Raine, A., & Gao, Y. (2009). Cross-Cultural Generalizability of the Reactive–Proactive Aggression Questionnaire (RPQ). Journal of Personality Assessment, 91(5), 473–479. http://doi.org/10.1080/00223890903088420
- Gansler, D. A., McLaughlin, N. C. R., Iguchi, L., Jerram, M., Moore, D. W., Bhadelia, R., & Fulwiler, C. (2009). A multivariate approach to aggression and the orbital frontal cortex in psychiatric patients. Psychiatry Research: Neuroimaging, 171(3), 145–154. http://doi.org/10.1016/j.pscychresns.2008.03.007
- Gao, Y., & Tang, S. (2013). Psychopathic personality and utilitarian moral judgment in college students. Journal of Criminal Justice, 41(5), 342–349.
- Gerra, G., Garofano, L., Pellegrini, C., Bosari, S., Zaimovic, A., Moi, G., ... Donnini, C. (2005). Allelic association of a dopamine transporter gene polymorphism with antisocial behaviour in heroin-dependent patients. Addiction Biology, 10(3), 275–281. http://doi.org/10.1080/13556210500223769
- Giegling, I., Hartmann, A. M., Möller, H.-J., & Rujescu, D. (2006). Anger- and aggression-related traits are associated with polymorphisms in the 5-HT-2A gene. Journal of Affective Disorders, 96(1–2), 75–81. http://doi.org/10.1016/j.jad.2006.05.016

- Gietl, A., Giegling, I., Hartmann, A. M., Schneider, B., Schnabel, A., Maurer, K., ...

 Rujescu, D. (2007). ABCG1 gene variants in suicidal behavior and aggressionrelated traits. European Neuropsychopharmacology, 17(6), 410–416.
- Glenn, A. L. (2011). The other allele: Exploring the long allele of the serotonin transporter gene as a potential risk factor for psychopathy: A review of the parallels in findings. Neuroscience & Biobehavioral Reviews, 35(3), 612–620. http://doi.org/10.1016/j.neubiorev.2010.07.005
- Goodwin, B. E., Sellbom, M., & Salekin, R. T. (2015). Elucidating the construct validity of the antisocial process screening device (APSD) in a sample of young adults.

 Journal of Psychopathology and Behavioral Assessment, 37(1), 1–11.
- Gotlib, I. H., Joormann, J., Minor, K. L., & Hallmayer, J. (2008). HPA axis reactivity: a mechanism underlying the associations among 5-HTTLPR, stress, and depression. Biological Psychiatry, 63(9), 847–851.
- Haberstick, B. C., Smolen, A., & Hewitt, J. K. (2006). Family-Based Association Test of the 5HTTLPR and Aggressive Behavior in a General Population Sample of Children. Biological Psychiatry, 59(9), 836–843.
 http://doi.org/10.1016/j.biopsych.2005.10.008
- Hariri, A. R., Mattay, V. S., Tessitore, A., Kolachana, B., Fera, F., Goldman, D., ... Weinberger, D. R. (2002). Serotonin transporter genetic variation and the response of the human amygdala. Science, 297(5580), 400–403.
- Hawes, D. J., Brennan, J., & Dadds, M. R. (2009). Cortisol, callous-unemotional traits, and pathways to antisocial behavior: Current Opinion in Psychiatry, 22(4), 357–362. http://doi.org/10.1097/YCO.0b013e32832bfa6d

- Holz, N., Boecker, R., Buchmann, A. F., Blomeyer, D., Baumeister, S., Hohmann, S., ...

 Laucht, M. (2016). Evidence for a Sex-Dependent MAOA× Childhood Stress

 Interaction in the Neural Circuitry of Aggression. Cerebral Cortex, 26(3), 904–914. http://doi.org/10.1093/cercor/bhu249
- Homberg, J. R., & Homberg, J. R. (2012). Serotonergic Modulation of Conditioned Fear, Serotonergic Modulation of Conditioned Fear. Scientifica, Scientifica, 2012, 2012, e821549. http://doi.org/10.6064/2012/821549, 10.6064/2012/821549
- Huang, Y., Cate, S. P., Battistuzzi, C., Oquendo, M. A., Brent, D., & Mann, J. J. (2004).
 An Association between a Functional Polymorphism in the Monoamine Oxidase
 A Gene Promoter, Impulsive Traits and Early Abuse Experiences.
 Neuropsychopharmacology, 29(8), 1498–1505.
 http://doi.org/10.1038/sj.npp.1300455
- Hubbard, J. A., Smithmyer, C. M., Ramsden, S. R., Parker, E. H., Flanagan, K. D.,
 Dearing, K. F., ... Simons, R. F. (2002). Observational, Physiological, and Self-Report Measures of Children's Anger: Relations to Reactive versus Proactive
 Aggression. Child Development, 73(4), 1101–1118.
- Jollant, F., Buresi, C., Guillaume, S., Jaussent, I., Bellivier, F., Leboyer, M., ... Courtet,
 P. (2007). The influence of four serotonin-related genes on decision-making in suicide attempters. American Journal of Medical Genetics Part B:
 Neuropsychiatric Genetics, 144B(5), 615–624.
 http://doi.org/10.1002/ajmg.b.30467
- Kempes, M., Matthys, W., De Vries, H., & Van Engeland, H. (2005). Reactive and proactive aggression in children A review of theory, findings and the relevance

- for child and adolescent psychiatry. European Child & Adolescent Psychiatry, 14(1), 11–19.
- Kiehl, K. A., Smith, A. M., Hare, R. D., Mendrek, A., Forster, B. B., Brink, J., & Liddle,
 P. F. (2001). Limbic abnormalities in affective processing by criminal
 psychopaths as revealed by functional magnetic resonance imaging. Biological
 Psychiatry, 50(9), 677–684.
- Kobak, R., Zajac, K., & Levine, S. (2009). Cortisol and antisocial behavior in early adolescence: The role of gender in an economically disadvantaged sample.
 Development and Psychopathology, 21(2), 579–91.
 http://doi.org/http://dx.doi.org/10.1017/S0954579409000315
- Kolla, N. J., Attard, S., Craig, G., Blackwood, N., & Hodgins, S. (2014). Monoamine oxidase A alleles in violent offenders with antisocial personality disorder: High activity associated with proactive aggression. Criminal Behaviour and Mental Health, 24(5), 368–372. http://doi.org/10.1002/cbm.1917
- Kuepper, Y., Alexander, N., Osinsky, R., Mueller, E., Schmitz, A., Netter, P., & Hennig, J. (2010). Aggression—Interactions of serotonin and testosterone in healthy men and women. Behavioural Brain Research, 206(1), 93–100.
 http://doi.org/10.1016/j.bbr.2009.09.006
- Kuepper, Y., Grant, P., Wielpuetz, C., & Hennig, J. (2013). MAOA-uVNTR genotype predicts interindividual differences in experimental aggressiveness as a function of the degree of provocation. Behavioural Brain Research, 247, 73–78. http://doi.org/10.1016/j.bbr.2013.03.002

- Langton, L., & Truman, J. (2014). Socio-emotional impact of violent crime. Special Report. Retrieved from http://www.bjs.gov/content/pub/pdf/sivc.pdf
- Latzman, R. D., Vaidya, J. G., Clark, L. A., & Watson, D. (2011). Components of disinhibition (vs. constraint) differentially predict aggression and alcohol use. European Journal of Personality, 25(6), 477–486.
- Laubscher, N., Odendaal, Z., Schneider, S., & Spies, J. J. (2013). A family study of reactive and proactive aggression in a South African population. Philos. Trans. Genet, 2, 1–16.
- Linnoila, M., Virkkunen, M., Scheinin, M., Nuutila, A., Rimon, R., & Goodwin, F. K. (1983). Low cerebrospinal fluid 5-hydroxyindoleacetic acid concentration differentiates impulsive from nonimpulsive violent behavior. Life Sciences, 33(26), 2609–2614.
- Loney, B. R., Butler, M. A., Lima, E. N., Counts, C. A., & Eckel, L. A. (2006). The relation between salivary cortisol, callous-unemotional traits, and conduct problems in an adolescent non-referred sample. Journal of Child Psychology and Psychiatry, 47(1), 30–36. http://doi.org/10.1111/j.1469-7610.2005.01444.x
- Loney, B. R., Frick, P. J., Clements, C. B., Ellis, M. L., & Kerlin, K. (2003). Callous-unemotional traits, impulsivity, and emotional processing in adolescents with antisocial behavior problems. Journal of Clinical Child and Adolescent Psychology, 32(1), 66–80.
- Lopez-Duran, N. L., Olson, S. L., Hajal, N. J., Felt, B. T., & Vazquez, D. M. (2008).

 Hypothalamic Pituitary Adrenal Axis Functioning in Reactive and Proactive

- Aggression in Children. Journal of Abnormal Child Psychology, 37(2), 169–182. http://doi.org/10.1007/s10802-008-9263-3
- Luppino, D., Moul, C., Hawes, D. J., Brennan, J., & Dadds, M. R. (2014). Association between a polymorphism of the vasopressin 1B receptor gene and aggression in children. Psychiatric Genetics, 24(5), 185–190.
- Manuck, S. B., Flory, J. D., Ferrell, R. E., Mann, J. J., & Muldoon, M. F. (2000). A regulatory polymorphism of the monoamine oxidase-A gene may be associated with variability in aggression, impulsivity, and central nervous system serotonergic responsivity. Psychiatry Research, 95(1), 9–23. http://doi.org/10.1016/S0165-1781(00)00162-1
- Marsee, M. A., & Frick, P. J. (2007). Exploring the Cognitive and Emotional Correlates to Proactive and Reactive Aggression in a Sample of Detained Girls. Journal of Abnormal Child Psychology, 35(6), 969–981. http://doi.org/10.1007/s10802-007-9147-y
- Marseille, R., Lee, R., & Coccaro, E. F. (2012). Inter-relationship between different platelet measures of 5-HT and their relationship to aggression in human subjects. Progress in Neuro-Psychopharmacology and Biological Psychiatry, 36(2), 277–281. http://doi.org/10.1016/j.pnpbp.2011.10.004
- Marsh, A. A., Finger, E. C., Mitchell, D. G., Reid, M. E., Sims, C., Kosson, D. S., ...
 Blair, R. J. R. (2008). Reduced amygdala response to fearful expressions in
 children and adolescents with callous-unemotional traits and disruptive behavior
 disorders. American Journal of Psychiatry. Retrieved from
 http://ajp.psychiatryonline.org/doi/abs/10.1176/appi.ajp.2007.07071145

- McBurnett K, Lahey BB, Rathouz PJ, & Loeber R. (2000). LOw salivary cortisol and persistent aggression in boys referred for disruptive behavior. Archives of General Psychiatry, 57(1), 38–43. http://doi.org/10.1001/archpsyc.57.1.38
- McCaffery, J. M., Bleil, M., Pogue-Geile, M. F., Ferrell, R. E., & Manuck, S. B. (2003).
 Allelic Variation in the Serotonin Transporter Gene-Linked Polymorphic Region
 (5-HTTLPR) and Cardiovascular Reactivity in Young Adult Male and Female
 Twins of European-American Descent: Psychosomatic Medicine, 65(5), 721–728.
 http://doi.org/10.1097/01.PSY.0000088585.67365.1D
- McCollister, K. E., French, M. T., & Fang, H. (2010). The cost of crime to society: New crime-specific estimates for policy and program evaluation. Drug and Alcohol Dependence, 108(1), 98–109.
- Meloy, J. R. (1992). The Psychopathic Mind: Origins, Dynamics, and Treatment.

 Rowman & Littlefield.
- Mercedes Perez-Rodriguez, M., Weinstein, S., New, A. S., Bevilacqua, L., Yuan, Q., Zhou, Z., ... Siever, L. J. (2010). Tryptophan-hydroxylase 2 haplotype association with borderline personality disorder and aggression in a sample of patients with personality disorders and healthy controls. Journal of Psychiatric Research, 44(15), 1075–1081. http://doi.org/10.1016/j.jpsychires.2010.03.014
- Meyer, J. H., Wilson, A. A., Rusjan, P., Clark, M., Houle, S., Woodside, S., ... Colleton, M. (2008). Serotonin'sub 2A' receptor binding potential in people with aggressive and violent behaviour. Journal of Psychiatry & Neuroscience: JPN, 33(6), 499–508.

- Meyer-Lindenberg, A., Buckholtz, J. W., Kolachana, B., Hariri, A. R., Pezawas, L.,
 Blasi, G., ... Weinberger, D. R. (2006). Neural Mechanisms of Genetic Risk for
 Impulsivity and Violence in Humans. Proceedings of the National Academy of
 Sciences of the United States of America, 103(16), 6269–6274.
- Miller, J. D., Lynam, D. R., & Jones, S. (2008). Externalizing behavior through the lens of the five-factor model: A focus on agreeableness and conscientiousness. Journal of Personality Assessment, 90(2), 158–164.
- Miller, R., Wankerl, M., Stalder, T., Kirschbaum, C., & Alexander, N. (2013). The serotonin transporter gene-linked polymorphic region (5-HTTLPR) and cortisol stress reactivity: a meta-analysis. Molecular Psychiatry, 18(9), 1018–1024. http://doi.org/10.1038/mp.2012.124
- Montoya, E. R., Terburg, D., Bos, P. A., & Honk, J. van. (2011). Testosterone, cortisol, and serotonin as key regulators of social aggression: A review and theoretical perspective. Motivation and Emotion, 36(1), 65–73. http://doi.org/10.1007/s11031-011-9264-3
- Murray-Close, D., Ostrov, J. M., Nelson, D. A., Crick, N. R., & Coccaro, E. F. (2010).
 Proactive, reactive, and romantic relational aggression in adulthood:
 Measurement, predictive validity, gender differences, and association with
 Intermittent Explosive Disorder. Journal of Psychiatric Research, 44(6), 393–404.
 http://doi.org/10.1016/j.jpsychires.2009.09.005
- Muthén, B., & Asparouhov, T. (2002). Latent variable analysis with categorical outcomes: Multiple-group and growth modeling in Mplus. Mplus Web Notes, 4(5), 1–22.

- Muthén & Muthén (2012). Mplus (Version 7.11). Los Angeles, CA: Muthén & Muthén.
- Muthén et al., (2007). Mplus: Statistical Analysis with Latent Variables (Version 4.21)[Computer software. Retrieved from http://citeseerx.ist.psu.edu/viewdoc/summary?doi=10.1.1.310.2841
- O'Leary, M. M., Taylor, J., & Eckel, L. (2010). Psychopathic personality traits and cortisol response to stress: The role of sex, type of stressor, and menstrual phase. Hormones and Behavior, 58(2), 250–256. http://doi.org/10.1016/j.yhbeh.2010.03.009
- Pardini, D. A., Raine, A., Erickson, K., & Loeber, R. (2014). Lower Amygdala Volume in Men is Associated with Childhood Aggression, Early Psychopathic Traits, and Future Violence. Biological Psychiatry, 75(1), 73–80.
 http://doi.org/10.1016/j.biopsych.2013.04.003
- Parsey, R. V., Oquendo, M. A., Simpson, N. R., Ogden, R. T., Van Heertum, R., Arango, V., & Mann, J. J. (2002). Effects of sex, age, and aggressive traits in man on brain serotonin 5-HT1A receptor binding potential measured by PET using [C-11]WAY-100635. Brain Research, 954(2), 173–182. http://doi.org/10.1016/S0006-8993(02)03243-2
- Passamonti, L., Fera, F., Magariello, A., Cerasa, A., Gioia, M. C., Muglia, M., ...

 Quattrone, A. (2006). Monoamine Oxidase-A Genetic Variations Influence Brain

 Activity Associated with Inhibitory Control: New Insight into the Neural

 Correlates of Impulsivity. Biological Psychiatry, 59(4), 334–340.

 http://doi.org/10.1016/j.biopsych.2005.07.027

- Pavlov, K. A., Chistiakov, D. A., & Chekhonin, V. P. (2012). Genetic determinants of aggression and impulsivity in humans. Journal of Applied Genetics, 53(1), 61–82.
- Perroud, N., Jaussent, I., Guillaume, S., Bellivier, F., Baud, P., Jollant, F., ... Courtet, P. (2010). COMT but not serotonin-related genes modulates the influence of childhood abuse on anger traits. Genes, Brain and Behavior, 9(2), 193–202. http://doi.org/10.1111/j.1601-183X.2009.00547.x
- Polman, H., Castro, B. O. de, Koops, W., Boxtel, H. W. van, & Merk, W. W. (2007). A Meta-Analysis of the Distinction between Reactive and Proactive Aggression in Children and Adolescents. Journal of Abnormal Child Psychology, 35(4), 522–535. http://doi.org/10.1007/s10802-007-9109-4
- Popma, A., Vermeiren, R., Geluk, C. A. M. L., Rinne, T., van den Brink, W., Knol, D. L., ... Doreleijers, T. A. H. (2007). Cortisol Moderates the Relationship between Testosterone and Aggression in Delinquent Male Adolescents. Biological Psychiatry, 61(3), 405–411. http://doi.org/10.1016/j.biopsych.2006.06.006
- Porter, S., & Woodworth, M. (2006). Psychopathy and Aggression. Handbook of Psychopathy, 481–494.
- Poulin, F., & Boivin, M. (2000). Reactive and proactive aggression: Evidence of a two-factor model. Psychological Assessment, 12(2), 115–122. http://doi.org/10.1037/1040-3590.12.2.115
- Poustka, L., Maras, A., Hohm, E., Fellinger, J., Holtmann, M., Banaschewski, T., ... Laucht, M. (2010). Negative association between plasma cortisol levels and aggression in a high-risk community sample of adolescents. Journal of Neural Transmission, 117(5), 621–627. http://doi.org/10.1007/s00702-010-0386-7

- Price, J. M., & Dodge, K. A. (1989). Reactive and Proactive Aggression in Childhood:

 Relations to Peer Status and Social Context Dimensions. Journal of Abnormal

 Child Psychology, 17(4), 455–472.
- Quiggle, N. L., Garber, J., Panak, W. F., & Dodge, K. A. (1992). Social information processing in aggressive and depressed children. Child Development, 63(6), 1305–1320.
- Raine, A., Dodge, K., Loeber, R., Gatzke-Kopp, L., Lynam, D., Reynolds, C., ... Liu, J. (2006). The Reactive–Proactive Aggression Questionnaire: Differential Correlates of Reactive and Proactive Aggression in Adolescent Boys. Aggressive Behavior, 32(2), 159–171. http://doi.org/10.1002/ab.20115
- Raine, A., Meloy, J. R., Bihrle, S., Stoddard, J., LaCasse, L., & Buchsbaum, M. S. (1998). Reduced prefrontal and increased subcortical brain functioning assessed using positron emission tomography in predatory and affective murderers.

 Behavioral Sciences & the Law, 16(3), 319–332.
- Rajender, S., Pandu, G., Sharma, J. D., Gandhi, K. P. C., Singh, L., & Thangaraj, K.
 (2008). Reduced CAG repeats length in androgen receptor gene is associated with violent criminal behavior. International Journal of Legal Medicine, 122(5), 367–372.
- Reif, A., Rösler, M., Freitag, C. M., Schneider, M., Eujen, A., Kissling, C., ... others. (2007). Nature and nurture predispose to violent behavior: serotonergic genes and adverse childhood environment. Neuropsychopharmacology, 32(11), 2375–2383.
- Retz, W., Retz-Junginger, P., Supprian, T., Thome, J., & Rösler, M. (2004). Association of serotonin transporter promoter gene polymorphism with violence: relation with

- personality disorders, impulsivity, and childhood ADHD psychopathology.

 Behavioral Sciences & the Law, 22(3), 415–425. http://doi.org/10.1002/bsl.589
- Rhee, S. H., & Waldman, I. D. (2002). Genetic and environmental influences on antisocial behavior: A meta-analysis of twin and adoption studies. Psychological Bulletin, 128(3), 490–529. http://doi.org/10.1037/0033-2909.128.3.490
- Rujescu, D., Giegling, I., Gietl, A., Hartmann, A. M., & Möller, H.-J. (2003). A functional single nucleotide polymorphism (V158M) in the COMT gene is associated with aggressive personality traits. Biological Psychiatry, 54(1), 34–39. http://doi.org/10.1016/S0006-3223(02)01831-0
- Sabol, S. Z., Hu, S., & Hamer, D. (1998). A functional polymorphism in the monoamine oxidase A gene promoter. Human Genetics, 103(3), 273–279.http://doi.org/10.1007/s004390050816
- Sadeh, N., Javdani, S., Jackson, J. J., Reynolds, E. K., Potenza, M. N., Gelernter, J., ...
 Verona, E. (2010). Serotonin transporter gene associations with psychopathic traits in youth vary as a function of socioeconomic resources. Journal of
 Abnormal Psychology, 119(3), 604.
- Salmivalli, C., & Nieminen, E. (2002). Proactive and reactive aggression among school bullies, victims, and bully-victims. Aggressive Behavior, 28(1), 30–44. http://doi.org/10.1002/ab.90004
- Scarpa, A., & Raine, A. (1997). Psychophysiology of anger and violent behavior.

 Psychiatric Clinics of North America, 20(2), 375–394.
- Schenk, A. M., Fremouw, W. J., & Keelan, C. M. (2013). Characteristics of college cyberbullies. Computers in Human Behavior, 29(6), 2320–2327.

- Schwartz, D., Dodge, K. A., Coie, J. D., Hubbard, J. A., Cillessen, A. H. N., Lemerise, E.
 A., & Bateman, H. (1998). Social-Cognitive and Behavioral Correlates of
 Aggression and Victimization in Boys' Play Groups. Journal of Abnormal Child
 Psychology, 26(6), 431–440.
- Sebastian, C. L., Roiser, J. P., Tan, G. C. Y., Viding, E., Wood, N. W., & Blakemore, S.-J. (2010). Effects of age and MAOA genotype on the neural processing of social rejection. Genes, Brain and Behavior, 9(6), 628–637. http://doi.org/10.1111/j.1601-183X.2010.00596.x
- Siever, L. J., Buchsbaum, M. S., New, A. S., Spiegel-Cohen, J., Wei, T., Hazlett, E. A., ... Mitropoulou, V. (1999). d,l-fenfluramine Response in Impulsive Personality Disorder Assessed with [18F]fluorodeoxyglucose Positron Emission Tomography. Neuropsychopharmacology, 20(5), 413–423. http://doi.org/10.1016/S0893-133X(98)00111-0
- Skondras, M., Markianos, M., Botsis, A., Bistolaki, E., & Christodoulou, G. (2004).

 Platelet monoamine oxidase activity and psychometric correlates in male violent offenders imprisoned for homicide or other violent acts. European Archives of Psychiatry and Clinical Neuroscience, 254(6), 380–386.

 http://doi.org/10.1007/s00406-004-0518-x
- Soderstrom, H., Blennow, K., Manhem, A., & Forsman, A. (2001). CSF studies in violent offenders¶I. 5-HIAA as a negative and HVA as a positive predictor of psychopathy. Journal of Neural Transmission, 108(7), 869–878.

 http://doi.org/10.1007/s007020170036

- Stadler, C., Kroeger, A., Weyers, P., Grasmann, D., Horschinek, M., Freitag, C., & Clement, H.-W. (2011). Cortisol reactivity in boys with attention-deficit/hyperactivity disorder and disruptive behavior problems: The impact of callous unemotional traits. Psychiatry Research, 187(1–2), 204–209. http://doi.org/10.1016/j.psychres.2010.05.004
- Stanford, M. S., Greve, K. W., & Gerstle, J. E. (1997). Neuropsychological correlates of self-reported impulsive aggression in a college sample. Personality and Individual Differences, 23(6), 961–965. http://doi.org/10.1016/S0191-8869(97)00120-7
- Sweet, R. A., Nimgaonkar, V. L., Kamboh, M. I., Lopez, O. L., Zhang, F., & DeKosky,
 S. T. (1998). Dopamine receptor genetic variation, psychosis, and aggression in
 Alzheimer disease. Archives of Neurology, 55(10), 1335–1340.
- Tosato, S., Bonetto, C., Di Forti, M., Collier, D., Cristofalo, D., Bertani, M., ... others. (2011). Effect of COMT genotype on aggressive behaviour in a community cohort of schizophrenic patients. Neuroscience Letters, 495(1), 17–21.
- Tuvblad, C., & Baker, L. A. (2011). Human aggression across the lifespan: genetic propensities and environmental moderators. Advances in Genetics, 75, 171.
- van de Giessen, E., Rosell, D. R., Thompson, J. L., Xu, X., Girgis, R. R., Ehrlich, Y., ... Siever, L. J. (2014). Serotonin transporter availability in impulsive aggressive personality disordered patients: A PET study with [11C]DASB. Journal of Psychiatric Research, 58, 147–154.

http://doi.org/10.1016/j.jpsychires.2014.07.025

- Vandenberg, R. J., & Lance, C. E. (2000). A review and synthesis of the measurement invariance literature: Suggestions, practices, and recommendations for organizational research. Organizational Research Methods, 3(1), 4–70.
- Van de Schoot, R., Lugtig, P., & Hox, J. (2012). A checklist for testing measurement invariance. European Journal of Developmental Psychology, 9(4), 486–492.
- Vitaro, F., Brendgen, M., & Tremblay, R. E. (2002). Reactively and proactively aggressive children: antecedent and subsequent characteristics. Journal of Child Psychology and Psychiatry, 43(4), 495–505. http://doi.org/10.1111/1469-7610.00040
- Vitaro, F., Gendreau, P. L., Tremblay, R. E., & Oligny, P. (1998). Reactive and Proactive Aggression Differentially Predict Later Conduct Problems. Journal of Child Psychology and Psychiatry, 39(3), 377–385. http://doi.org/10.1111/1469-7610.00333
- Witte, A. V., Flöel, A., Stein, P., Savli, M., Mien, L.-K., Wadsak, W., ... Lanzenberger, R. (2009). Aggression is related to frontal serotonin-1A receptor distribution as revealed by PET in healthy subjects. Human Brain Mapping, 30(8), 2558–2570. http://doi.org/10.1002/hbm.20687
- Yildirim, B. O., & Derksen, J. J. (2013). Systematic review, structural analysis, and new theoretical perspectives on the role of serotonin and associated genes in the etiology of psychopathy and sociopathy. Neuroscience & Biobehavioral Reviews, 37(7), 1254–1296.
- Young, S. E., Smolen, A., Corley, R. P., Krauter, K. S., DeFries, J. C., Crowley, T. J., & Hewitt, J. K. (2002). Dopamine transporter polymorphism associated with

- externalizing behavior problems in children. American Journal of Medical Genetics, 114(2), 144–149.
- Zai, C. C., Ehtesham, S., Choi, E., Nowrouzi, B., De Luca, V., Stankovich, L., ... others.(2012). Dopaminergic system genes in childhood aggression: Possible role for DRD2. The World Journal of Biological Psychiatry, 13(1), 65–74.
- Zhang, X., Beaulieu, J.-M., Sotnikova, T. D., Gainetdinov, R. R., & Caron, M. G. (2004).

 Tryptophan Hydroxylase-2 Controls Brain Serotonin Synthesis. Science,

 305(5681), 217–217. http://doi.org/10.1126/science.1097540
- Zouk, H., McGirr, A., Lebel, V., Benkelfat, C., Rouleau, G., & Turecki, G. (2007). The effect of genetic variation of the serotonin 1B receptor gene on impulsive aggressive behavior and suicide. American Journal of Medical Genetics Part B: Neuropsychiatric Genetics, 144B(8), 996–1002.
 http://doi.org/10.1002/ajmg.b.30521

APPENDIX A

Reactive-Proactive Aggression Questionnaire

There are times when most of us feel angry, or have done things we should not have done. Don't spend a lot of time thinking about the items – just give your first response.

How often have you...

- -Yelled at others when they have annoyed you
- -Had fights with others to show who was on top
- -Reacted angrily when provoked by others
- -Taken things from other students
- -Had temper tantrums
- -Vandalized something for fun
- -Damaged things because you felt mad
- -Had a gang fight to be cool
- -Gotten angry when frustrated
- -Hurt others to win a game
- -Become angry or mad when you lost a game
- -Used physical force to get others to do what you want
- -Threatened and bullied someone
- -Gotten angry when others threatened you
- -Used force to obtain money or things from others
- -Damaged things because you felt angry
- -Made obscene phone calls for fun
- -Felt better after hitting or yelling at someone
- -Threatened or forced someone to have
- -Gotten angry or mad when you lost a game
- -Hit others to defend yourself
- -Carried a weapon to use in a fight
- -Gotten angry or mad or hit others when teased

VITA

EDUCATION

Masters of Arts in Criminal Justice and Criminology at Sam Houston State University, Huntsville, Texas. Expected graduation December 2016.

Bachelor of Arts in Psychology at Miami University, Oxford, Ohio. May 2012.

PRESENTATIONS

Cooke, E. (November, 2015). "Validation of An Aggression Measurement Tool Across Gender and Ethnicity." American Society of Criminology. Washington, D.C.

ACADEMIC AWARDS

Graduate Research Assistantship, College of Criminal Justice, Sam Houston State University, 2014-present.

Employee Service Leadership Award, Miami University, 2012.

ACADEMIC EMPLOYMENT

August 2014 to present: Graduate Research Assistant for the College of Criminal Justice, Sam Houston State University.

August 2011 to May 2012: Undergraduate Research Assistant for the Department of Psychology, Miami University. Under the Direction of Dr. Wolfe.

ORGANIZATIONAL MEMBERSHIP

2015 to present: Criminal Justice Graduate Student Organization; Secretary

2014 to present: Criminal Justice Graduate Student Organization

2011 to 2012: Student Ambassador, Miami University

PROFESSIONAL MEMBERSHIP

Academy of Criminal Justice Sciences

American Society of Criminology